

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Robotti, Cesare

Working Paper The price of inflation and foreign exchange risk in international equity markets

Working Paper, No. 2001-26

Provided in Cooperation with: Federal Reserve Bank of Atlanta

Suggested Citation: Robotti, Cesare (2001) : The price of inflation and foreign exchange risk in international equity markets, Working Paper, No. 2001-26, Federal Reserve Bank of Atlanta, Atlanta, GA

This Version is available at: https://hdl.handle.net/10419/100831

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



FEDERAL RESERVE BANK of ATLANTA

The Price of Inflation and Foreign Exchange Risk in International Equity Markets

Cesare Robotti

Working Paper 2001-26 November 2001

Working Paper Series

Federal Reserve Bank of Atlanta Working Paper 2001-26 November 2001

The Price of Inflation and Foreign Exchange Risk in International Equity Markets

Cesare Robotti, Federal Reserve Bank of Atlanta

Abstract: In this paper the author formulates and tests an international intertemporal capital asset pricing model in the presence of deviations from purchasing power parity (II-CAPM [PPP]). He finds evidence in favor of at least mild segmentation of international equity markets in which only global market risk appears to be priced. When using the Hansen & Jagannathan (1991, 1997) variance bounds and distance measures as testing devices, the author finds that, while all international asset pricing models are formally rejected by the data, their pricing implications are substantially different. The superior performance of the II-CAPM (PPP) is mainly attributable to significant hedging against inflation risk.

JEL classification: G12, G15

Key words: international intertemporal capital asset pricing model, purchasing power parity, hedging demands

The author thanks Pierluigi Balduzzi, Arthur Lewbel, Shijun Liu, and seminar participants at Boston College, the University of California at Irvine, and the 2000 Meetings of the Financial Management Association for useful comments. The views expressed here are the author's and not necessarily those of the Federal Reserve Bank of Atlanta or the Federal Reserve System. Any remaining errors are the author's responsibility.

Please address questions regarding content to Cesare Robotti, Financial Economist, Research Department, Federal Reserve Bank of Atlanta, 1000 Peachtree Street, N.E., Atlanta, Georgia 30309-4470, 404-498-8543, 404-498-8810 (fax), cesare.robotti@atl.frb.org.

The full text of Federal Reserve Bank of Atlanta working papers, including revised versions, is available on the Atlanta Fed's Web site at http://www.frbatlanta.org/publica/work_papers/index.html. To receive notification about new papers, please use the on-line publications order form, or contact the Public Affairs Department, Federal Reserve Bank of Atlanta, 1000 Peachtree Street, N.E., Atlanta, Georgia 30309-4470, 404-498-8020.

Introduction

In recent years, several papers have tested the international version of the Capital Asset Pricing Model (I–CAPM). Some of these papers have found that in addition to compensation for global market risk, investors require compensation for inflation and foreign exchange risk. However, the evidence on the ability of different specifications of the I–CAPM to correctly price international asset returns is mixed. Dumas and Solnik (1995) and De Santis and Gérard (1998) cannot reject a specification of the I–CAPM that includes global market risk and foreign exchange risk. Their analysis is consistent with the hypothesis of integration of international equity and foreign exchange markets. On the contrary, Vassalou (2000) rejects the adequacy of several nested versions of the I–CAPM with foreign exchange and inflation risks to explain most of the cross–sectional variation in stock excess returns. Her work seems to support the hypothesis of at least mild segmentation in international stock markets.

Moreover, the evidence regarding the size and significance of the economic risk premia is less than conclusive. While there is some evidence of time variation in the premia, the patterns of time variation are somewhat unclear. For example, De Santis and Gérard (1997,1998) find that exposure to global market risk commands a positive and significant average risk premium only when a conditional, fully parametric specification of the I-CAPM with timevarying price of risk is assumed. Similarly, Ferson and Harvey (1993) find a positive and significant risk premium for exposure to global market risk in a time-varying multi-beta international asset pricing model. Dumas and Solnik (1995) provide evidence of time variation in the global market risk premium, but do not report size and statistical significance of the average conditional market premium. Vassalou (2000) does not comment on the sign and the significance of the unconditional global market premium. The evidence on the premia associated with foreign exchange risk and inflation risk is also unclear. For example, De Santis and Gérard (1998) find support for a specification of the conditional I–CAPM that includes both global market risk and foreign exchange risk under the assumption of time variation in all prices of risk. Their results are globally consistent with the findings of Dumas and Solnik (1995). Even if both studies assume time-varying prices of risk, the main difference between the two approaches is that Dumas and Solnik (1995) test an unconditional version of the conditional I–CAPM using the generalized method of moments, while De Santis and Gérard (1998) explicitly model time variation in the first two conditional moments using VGARCH specifications. On the other hand, Vassalou (2000) finds strong support for versions of the I–CAPM that take inflation risk into account, mixed evidence for versions of the I–CAPM with foreign exchange risk, and no support for international asset pricing models where both foreign exchange premia and inflation premia are jointly estimated. In addition, the magnitudes of the estimated risk premia change substantially from one study to the other.

Two main issues emerge from the previous discussion. First, it is hard to reconcile home bias in international equity and foreign exchange markets with the findings that stock markets are perfectly integrated and that the I-CAPM holds. One of the reasons why the I-CAPM holds in some studies and not in others might be that the testing procedure used crucially depends on the level of aggregation of the equity portfolios used in the analysis. For example, Vassalou (2000) considers not only cross-country variation but also within-country variation in equity returns in order to control for home bias. On the other hand, Dumas and Solnik (1995) and De Santis and Gérard (1998) use aggregated stock return data from Morgan Stanley Capital International (MSCI). In the effort to control for home bias, the authors include in the analysis idiosyncratic risks and country-specific effects. Nonetheless, De Santis and Gérard (1997,1998) reach a quite puzzling result. They find that when the restriction of non-negativity of the conditional global market risk premium is imposed, there is some residual predictability in stock excess returns that can be explained by idiosyncratic risk and country-specific effects. On the other hand, when the restriction of non-negativity is relaxed, this residual predictability disappears and they cannot reject the hypothesis of integration of international stock markets. Second, the patterns of time variation in the first two conditional moments of excess asset returns and in the prices of risk are not clear. The I-CAPM in its unconditional version does not seem to hold.¹ It also does not seem to hold in its conditional version unless all prices of risk are assumed to be time-varying.² I try to

¹See, for example, Solnik (1974), and Stehle (1977).

²See, for example, Hodrick (1981), De Santis and Gérard (1998), Dumas and Solnik (1995), and Bekaert and Harvey (1995).

clarify the above-mentioned two issues by separating the problem of economic risk premia estimation from the problem of testing intertemporal international asset pricing models and by implementing a convenient new approach to the estimation of economic risk premia. I also implement a comprehensive set of tests of asset pricing models with foreign exchange and inflation risks.

This paper contributes to the existing literature in three different ways. First, I use a new methodology to estimate the inflation and foreign exchange risk premia which is based on the minimum-variance stochastic discount factors of Hansen and Jagannathan (1991) (HJ). This pricing kernel prices, by construction, the asset returns under consideration, and has the minimum variance among all kernels consistent with asset returns. The economic risk premia that I estimate are those assigned by the minimum-variance kernel. This approach has several advantages on the traditional methods:

- I do not need to specify all relevant sources of risk; i.e., the estimate of a risk premium does not change depending on which other sources of uncertainty are simultaneously considered.
- The estimates are robust to the form of the pricing kernel. Any admissible pricing kernel has the same price implications and, hence, assigns the same risk premia as the HJ minimum-variance kernel.
- The estimation procedure is quite simple. I estimate the parameters of the minimumvariance kernel and the covariance of the kernel with the different sources of uncertainty. By comparison, studies based on multi-beta models require the estimation of linear factor models for each of the securities, and then the estimation of the risk premia.
- I show that when the non-negativity restriction on the pricing kernel is not imposed, the proposed economic risk premia reduce to the expected cash flows on the portfolios that best hedge the risk factors, financed at the riskless rate. This result makes it clear that the risk premia estimates crucially depend on the hedging portfolios composition, and hence on the assets available for investment.

Second, I model the time variation of economic risk premia by specifying a pricing kernel linear in the set of instruments used in the analysis. Finally, I use a new methodology to test the international CAPM, which is based on the construction of portfolios mimicking the behavior of global market risk, foreign exchange, and inflation risks. Specifically, I test the International Static CAPM (IS-CAPM), the International CAPM with demands hedging against foreign exchange and inflation risks (I-CAPM), and the International Intertemporal CAPM (II-CAPM) by using a variety of diagnostics: the standard J-test of overidentifying restrictions of Hansen and Singleton (1982); the Hansen–Jagannathan (1991) variance bound; and the Hansen–Jagannathan (1997) distance measure. I am not aware of any formal test of the I–CAPM with demands hedging against variations in the investment opportunity set of an international investor in presence of deviations from Purchasing Power Parity (PPP) and currency risk.³

The main results of the paper can be summarized as follows: First, I show that only global market risk commands a highly positive and significant unconditional risk premium as indicated by its relative Sharpe ratio. Inflation risk does not seem to be priced both unconditionally and conditionally, while global market and exchange risks exhibit interesting patterns of time variation. Second, none of the international asset pricing models are rejected by the data when, following Hansen and Singleton (1982), I perform tests of the overidentifying restrictions that they impose. On the contrary, testing the same specifications with the HJ variance bounds and distance measures leads to a strong rejection of all models.⁴ Hence, even if all international asset pricing models deliver different pricing implications, none of them seems to hold when the testing procedure adopted is stringent enough. Large and statistically significant inflation-hedging demands explain the better pricing implications delivered by the II-CAPM in presence of deviations from PPP. Interestingly, this analysis is able to rationalize recent findings in the international asset pricing literature and, at the same time, shed some light on the controversial home bias puzzle. Even if I cannot explain

³Hodrick, Ng, and Sengmüller (1999) test an international version of Campbell's (1996) intertemporal CAPM, but do not consider deviations from PPP.

⁴See Zhang (2001) for an evaluation of domestic and international asset pricing models using the HJ (1997) distance measure.

home bias in international equity markets, I can argue that my results are consistent with the hypothesis of at least mild segmentation of international stock markets.

The remainder of this paper is organized as follows: Section I discusses the economic risk premia assigned by the minimum-variance admissible kernel and explicit international asset pricing models;⁵ Section II illustrates the methods used for estimation and testing; Section III describes the data; Section IV identifies the investment opportunity set of an investor; Section V presents the results of the risk premia estimation and relative Sharpe ratios; Section VI presents the results of the tests of international asset pricing models and empirically identifies the size of the hedging demands; and Section VII concludes.

I. Methodology: Economic Risk Premia and International Asset Pricing Models

This section derives a general pricing result for nominal asset returns denominated in terms of a reference currency. Namely, I show that the expected nominal returns in excess of the nominal risk-free rate of the reference country are directly related to the covariance of a nominal pricing kernel, scaled by the risk-free rate, with nominal asset returns. I then consider economic variables that are believed to be relevant for international asset prices: the rate of return on the world market portfolio; the rate of appreciation of the reference currency relative to the other currencies; and the rates of inflation in the different countries. If I can construct portfolios that exactly replicate the conditional variability of these variables, then the cross moments of the scaled (or normalized) pricing kernel translate into risk premia. Specifically, these cross moments reduce to the expected nominal cash flows on zero-netinvestment positions long in the mimicking portfolio and short in the riskless asset. If the nominal (normalized) pricing kernel is linear in the economic variables described above, as is often assumed in international asset pricing models, then the risk premia also enter familiar "beta" pricing results. In addition, I show how to test several specifications of the

⁵The analysis of risk premia mainly follows Balduzzi & Kallal (1997) and Balduzzi & Robotti (2001).

international capital asset pricing model using a variety of diagnostics.

A. A General International Asset Pricing Result

Assume there are L+1 countries and a set of N = n+1 assets, other than the measurementcurrency nominally risk-free asset. These include n risky assets, or portfolios of risky assets, and the world portfolio of risky assets.

Consider now the $N \times 1$ vector of (gross) nominal returns on the N assets, r. From now on, I shall assume all returns to be denominated in terms of the reference currency. By the law of one price, I have

$$\mathbf{E}_t(m_{t+1}\mathbf{r}_{t+1}) = \mathbf{1} \tag{1}$$

for some admissible nominal pricing kernel m,⁶ where **1** is an $N \times 1$ vector of ones. Note that the restriction stated in equation (1) prevents arbitrage in the security markets, but **not** in the commodity markets where Purchasing Power Parity (PPP) may not hold.

Let r_{ft} denote the nominally risk-free rate of the reference country. I then have

$$E_t(m_{t+1}r_{ft}) = 1. (2)$$

It is convenient to perform the analysis that follows in terms of the pricing kernel scaled by the risk-free rate. That is, I use the normalized pricing kernel, $m_{t+1}r_{ft} \equiv q_{t+1}$, where $E_t(q_{t+1}) =$ 1. As a consequence, I circumvent the problem of explicitly modeling the conditional mean of the pricing kernel m.

Using equations (1) and (2), I obtain the familiar orthogonality conditions

$$E_t[q_{t+1}(\mathbf{r}_{t+1} - \mathbf{1}\mathbf{r}_{ft})] = \mathbf{0} .$$
(3)

Rearranging, I obtain

$$E_t(q_{t+1})[E_t(\mathbf{r}_{t+1}) - \mathbf{1}\mathbf{r}_{ft}] = E_t(\mathbf{r}_{t+1}) - \mathbf{1}\mathbf{r}_{ft} = -Cov_t(\mathbf{q}_{t+1}, \mathbf{r}_{t+1}).$$
(4)

⁶See Harrison and Kreps (1979). The set of pricing kernels can be interpreted as the set of nominal intertemporal marginal rates of substitution compatible with the distribution of returns.

The conditional risk premium on any asset equals the opposite of the conditional covariance between the normalized pricing kernel and the asset return.

B. Economic Variables and Exact Mimicking Portfolios

Consider now a $K \times 1$ vector of economic variables **y**. Without loss of generality, I assume these variables to have constant zero mean and unit variance: i.e., $E_t(y_{k,t+1}) = 0$ and $E_t(y_{k,t+1}^2) = 1$, for k = 1, ..., K.

The reason for assuming a constant zero mean is that security markets only price unanticipated variability; the relevant economic variables are, in fact, innovations. Finally, the reason for scaling the risk factors to have constant unit variance is that I can compare the "prices" attached to the economic variables without having to worry about differences in variability.

Assume now that there exist portfolios that exactly mimic the behavior of the risk factors. Let $\mathbf{r}_{y,t+1} \equiv \mathbf{y}_{t+1}$, denote the $K \times 1$ vector of mimicking portfolio returns. According to (4), I have

$$\mathbf{E}_t(\mathbf{r}_{y,t+1}) - r_{ft}\mathbf{1} = -\mathbf{E}_t[\mathbf{y}_{t+1}(q_{t+1}-1)] \equiv \boldsymbol{\lambda}_{yt} , \qquad (5)$$

which is the vector of conditional risk premia on the corresponding economic variables. Hence, the conditional cross moments between the normalized pricing kernel q and the economic variable y_k equals, with the opposite sign, the conditional risk premium on the variable. Assuming stationarity and using the law of iterated expectations, I have

$$\boldsymbol{\lambda}_y \equiv \mathrm{E}(\boldsymbol{\lambda}_{yt}) \equiv -\mathrm{E}[(q_{t+1}-1)\mathbf{y}_{t+1}]$$
,

which is the vector of unconditional or mean risk premia on \mathbf{y} . The unconditional cross moment between the normalized pricing kernel q and the economic variable y_k equals, with the opposite sign, the mean risk premium on the variable.

C. The Minimum-Variance Kernel

Using the definition of the normalized pricing kernel q, I can rewrite equations (1) and (2) as

$$\mathbf{E}_t(q_{t+1}\mathbf{r}_{t+1}) = r_{ft}\mathbf{1} \tag{6}$$

and

$$E_t(q_{t+1}) = 1$$
. (7)

Following HJ, I can construct an admissible (normalized) pricing kernel, i.e., a random variable that satisfies equations (6) and (7), that is linear in \mathbf{r}_{t+1} : $q_{t+1}^{\star} \equiv \alpha_{0t} + \mathbf{r}_{t+1}^{\top} \boldsymbol{\alpha}_t$, where α_{0t} is a scalar and $\boldsymbol{\alpha}_t$ is an $N \times 1$ coefficient vector. I then have

$$E_t(q_{t+1})E_t(\mathbf{r}_{t+1}) + Cov_t(q_{t+1}, \mathbf{r}_{t+1}) = r_{ft}\mathbf{1}.$$
(8)

Using (6) and $q_{t+1}^{\star} \equiv \alpha_{0t} + \mathbf{r}_{t+1}^{\top} \boldsymbol{\alpha}_{t}$, I have

$$-\Sigma_{rrt}\boldsymbol{\alpha}_t = \mathbf{E}_t(\mathbf{r}_{t+1}) - r_{ft}\mathbf{1} , \qquad (9)$$

where Σ_{rrt} is the conditional covariance matrix of risky asset returns (which I assume to be invertible). I obtain

$$\boldsymbol{\alpha}_t = -\boldsymbol{\Sigma}_{rrt}^{-1} [\mathbf{E}_t(\mathbf{r}_{t+1}) - r_{ft} \mathbf{1}]$$
(10)

while

$$\alpha_{0t} = 1 - \mathbf{E}_t(\mathbf{r}_{t+1})^\top \boldsymbol{\alpha}_t \ . \tag{11}$$

Using the results in equations (10) and (11), I have

$$q_{t+1}^{\star} = 1 - [\mathbf{r}_{t+1} - \mathbf{E}_t(\mathbf{r}_{t+1})]^{\top} \boldsymbol{\Sigma}_{rrt}^{-1} [\mathbf{E}_t(\mathbf{r}_{t+1}) - r_{ft} \mathbf{1}] .$$
(12)

This minimum-variance normalized pricing kernel, q_{t+1}^{\star} , has several properties worth noting. First, the vector $\boldsymbol{\alpha}_t$ is proportional to the vector of portfolio weights of the tangency portfolio obtained from the risky security returns \mathbf{r}_{t+1} .⁷ Hence, q_{t+1}^{\star} is perfectly negatively correlated with the rate of return on the tangency portfolio, $r_{\tau,t+1}$. Another important property is that the conditional variance of q_{t+1}^{\star} equals the squared conditional Sharpe ratio of the tangency portfolio, $S_{\tau t}$. I have

$$[\operatorname{Cov}_t(q_{t+1}^{\star}, r_{\tau, t+1})]^2 = \operatorname{Var}_t(r_{\tau, t+1}) \operatorname{Var}_t(q_{t+1}^{\star}) .$$
(13)

Since q_{t+1}^{\star} correctly prices all the securities under consideration, it also correctly prices the tangency portfolio, and I have $-\text{Cov}_t(q_{t+1}^{\star}, r_{\tau,t+1}) = \text{E}_t(r_{\tau,t+1}) - r_{ft}$. Using this result and rearranging equation (13) above, I obtain

$$\left[\frac{\mathbf{E}_t(r_{\tau,t+1}) - r_{ft}}{\sqrt{\mathrm{Var}_t(r_{\tau,t+1})}}\right]^2 \equiv \mathbf{S}_{\tau t}^2 = \mathrm{Var}_t(q_{t+1}^\star) \ . \tag{14}$$

Finally, since $\operatorname{Var}_t(q_{t+1}^{\star}) = \operatorname{E}_t[(q_{t+1}^{\star} - 1)^2]$, then $\operatorname{Var}(q_{t+1}^{\star}) = \operatorname{E}[(q_{t+1}^{\star} - 1)^2]$, and

$$\operatorname{Var}(q_{t+1}^{\star}) = \operatorname{E}(\operatorname{S}_{\tau t}^{2}) . \tag{15}$$

Hence, the unconditional variance of q_{t+1}^{\star} equals the average squared Sharpe ratio of the tangency portfolio.

D. Hedging Portfolios and Risk Premia

Now consider the conditional projection of the risk variable $y_{k,t+1}$ on a constant and \mathbf{r}_{t+1} $(y_{k,t+1}^{\star} \equiv \alpha_{0y_k t} + \mathbf{r}_{t+1}^{\top} \boldsymbol{\alpha}_{y_k t}, \text{ where } \alpha_{0y_k t} \text{ is a scalar and } \boldsymbol{\alpha}_{y_k t} \text{ is an } N \times 1 \text{ coefficient vector}).$ Let $\lambda_{kt}^{\star} \equiv -\text{Cov}_t(q_{t+1}y_{k,t+1}^{\star}) = -\text{Cov}_t(q_{t+1}^{\star}y_{k,t+1}^{\star}).$ Since q_{t+1}^{\star} satisfies the conditional moment restriction (6) by construction, I have

$$\lambda_{kt}^{\star} = -\operatorname{Cov}_t(q_{t+1}^{\star}, y_{k,t+1}^{\star}) = -\operatorname{Cov}_t(q_{t+1}^{\star}, \boldsymbol{\alpha}_{y_k t}^{\top} \mathbf{r}_{t+1})$$
$$= -\boldsymbol{\alpha}_{y_k t}^{\top} \operatorname{Cov}_t(q_{t+1}^{\star}, \mathbf{r}_{t+1}) = \boldsymbol{\alpha}_{y_k t}^{\top} [\operatorname{E}_t(\mathbf{r}_{t+1}) - r_{ft} \mathbf{1}] , \qquad (16)$$

which is the mean cash flow generated by the portfolio hedging the risk variable y_k financed at the riskless rate. In other words, λ_{kt}^{\star} is the risk premium on the hedging portfolio for the

⁷For example, see Ingersoll (1987), p. 89.

risk variable $y_{k,t+1}$. This implies that λ_{kt}^* does **not** depend on the choice of the (normalized) pricing kernel, but only on the asset returns under scrutiny. If the conditional volatility of $y_{k,t+1}$ equals one and the factor is exactly replicated by asset returns, then λ_{kt}^* is also the conditional Sharpe ratio on the exact hedging portfolio.

The hedging portfolios defined here are analogous to the "economic tracking portfolios" of Lamont (2001). The main difference in his approach is that the portfolios are constructed to track changes in expectations of future realizations of the economic variables. Instead, the proposed hedging portfolios are designed to track the contemporaneous realizations of economic variables.

The composition of the hedging portfolios is worth further discussion. First, the hedging portfolios contain allocations to the riskless asset in the amount α_{0y_kt}/r_{ft} . (The rate of return on the hedging portfolio contains a constant component, α_{0y_kt} , resulting from the investment in the riskless asset.) Second, the hedging portfolio quantities do not sum up to one. In order to have a "true" hedging portfolio I have to scale the coefficients α_{0y_kt} and α_{y_kt} by $\alpha_{0y_kt}/r_{ft} + \mathbf{1}^{\top} \boldsymbol{\alpha}_{y_kt}$. Finally, the composition of the hedging portfolios corresponds to the coefficients of a regression of the risk factors on the asset returns

$$[\alpha_{0y_kt}, \boldsymbol{\alpha}_{y_kt}^{\top}]^{\top} = [\mathbf{E}_t(\mathbf{r}_{t+1}\mathbf{r}_{t+1}^{\top})]^{-1} \mathbf{E}_t(\mathbf{r}_{t+1}y_{k,t+1}) .$$
(17)

Hence, the hedging portfolios are the discrete-time counterparts of the portfolios held by a dynamic portfolio optimizer to hedge against changes in the investment-opportunity set.⁸

Two properties relating to the estimation of the risk premia are also worth noting. First, while the conditional expected excess cash flow on a mimicking portfolio equals the conditional covariance between the factor and the minimum-variance kernel, the realized excess cash flow on the mimicking portfolio, $\boldsymbol{\alpha}_{y_k t}^{\top}(\mathbf{r}_{t+1}-r_{ft}\mathbf{1})$, in general differs from $-(q_{t+1}^{\star}-1)y_{k,t+1}$. In fact,

$$\mathbf{E}_{t}(y_{k,t+1}\mathbf{r}_{t+1}^{\top})\mathbf{E}_{t}(\mathbf{r}_{t+1}\mathbf{r}_{t+1}^{\top})^{-1}(\mathbf{r}_{t+1}-r_{f}\mathbf{1}) \neq [\mathbf{r}_{t+1}-\mathbf{E}_{t}(\mathbf{r}_{t+1})]^{\top}\boldsymbol{\Sigma}_{rrt}^{-1}[\mathbf{E}_{t}(\mathbf{r}_{t+1})-r_{ft}\mathbf{1}]y_{k,t+1}.$$
 (18)

This observation is important because the estimates of the risk premia may differ in small

⁸See Merton (1973).

samples, depending on which approach is used, and the precision of the estimates may also differ. Second, the conditional Sharpe ratios of the hedging portfolios, $S_{y_k^{\star t}}$, depend on how closely a risk variable is replicated. Since the replication is not perfect, in general $\operatorname{Var}_t(y_{k,t+1}^{\star}) < \operatorname{Var}_t(y_{k,t+1})$. This means that

$$S_{y_k^{\star t}}^2 = \frac{(\lambda_{kt}^{\star})^2}{\operatorname{Var}_t(y_{k,t+1}^{\star})} > \frac{(\lambda_{kt}^{\star})^2}{\operatorname{Var}_t(y_{k,t+1})}$$

This observation is important because a hedging portfolio might receive a "small" risk premium and yet command a high Sharpe ratio. This is because it only captures a small fraction of the variability of a risk variable.

Appendix A relates my approach to the estimation of economic risk premia to two alternative approaches widely considered in the literature: 1) Principal components approach; and 2) Multi-beta models approach. The first approach assumes that a number of unobservable factors drive the variation in asset returns. The realizations of these factors, while not directly observable, can be inferred from the statistical properties of asset returns. The second approach explicitly identifies the factors with observable macro-economic variables, and assumes the pricing kernel and/or asset returns to be linear in the factors.

E. Hedging Portfolios and Linear Kernels

In this section, I describe the link between hedging portfolios and linear kernels. This link is then used in Section F to formulate and test competitive international asset pricing models. Without loss of generality, I assume $E_t(\mathbf{y}_{t+1}) = \mathbf{0}$ and $E_t(\mathbf{y}_{t+1}\mathbf{y}_{t+1}^{\top}) \equiv \boldsymbol{\Sigma}_{yyt} = \mathbf{I}$. Consider now an admissible kernel $q_{y,t+1}$ which is linear in the risk variables

$$q_{y,t+1} = 1 - \mathbf{y}_{t+1}^{\top} \boldsymbol{\lambda}_t . \tag{19}$$

Equation (4) above becomes

$$\mathbf{E}_t(\mathbf{r}_{t+1}) - \mathbf{1}r_{ft} = \boldsymbol{\beta}_t \boldsymbol{\lambda}_t , \qquad (20)$$

where $\boldsymbol{\beta}_t = \mathbf{E}_t(\mathbf{r}_{t+1}\mathbf{y}_{t+1}^{\top}) \equiv \boldsymbol{\Sigma}_{ryt}$. This implies that returns satisfy the linear factor model

$$\mathbf{r}_{t+1} = \mathbf{1}r_{ft} + \boldsymbol{\beta}_t \boldsymbol{\lambda}_t + \boldsymbol{\beta}_t \mathbf{y}_{t+1} + \boldsymbol{\epsilon}_{t+1} , \qquad (21)$$

where λ_t is a $K \times 1$ vector of conditional risk premia and ϵ_{t+1} is an $N \times 1$ vector of disturbances orthogonal to \mathbf{y}_{t+1} .

Now consider the projection of the risk variables onto the span of asset returns augmented of a unit-constant. I have $\mathbf{y}_{t+1}^{\star} \equiv \boldsymbol{\alpha}_{0yt} + \boldsymbol{\alpha}_{yt}\mathbf{r}_{t+1}^{\top}$, where $\boldsymbol{\alpha}_{0yt}$ is a $K \times 1$ vector and $\boldsymbol{\alpha}_{yt}$ is a $K \times N$ coefficient matrix. I have

$$\boldsymbol{\alpha}_{yt} = \boldsymbol{\Sigma}_{yrt} \boldsymbol{\Sigma}_{rrt}^{-1} , \qquad (22)$$

$$\boldsymbol{\alpha}_{0yt} = -\boldsymbol{\alpha}_{yt} \mathbf{E}_t(\mathbf{r}_{t+1}) , \qquad (23)$$

and

$$\mathbf{y}_{t+1}^{\star} \equiv \boldsymbol{\alpha}_{yt}[\mathbf{r}_{t+1} - \mathbf{E}_t(\mathbf{r}_{t+1})] .$$
⁽²⁴⁾

If, in the admissible linear pricing kernel q_{t+1} , I replace \mathbf{y}_{t+1} with \mathbf{y}_{t+1}^{\star} , the pricing result (20) does not change. In other words, the projection of $q_{y,t+1}$ onto the augmented span of asset returns, $q_{y^{\star},t+1} \equiv (\mathbf{y}_{t+1}^{\star})^{\top} \boldsymbol{\lambda}_{t}$, is also an admissible pricing kernel.

Consider now the minimum-variance kernel constructed based on the cash flows of the hedging portfolios $q_{y^{\star},t+1}^{\star}$. This is the kernel with minimum variance that correctly prices the hedging portfolios. I have

$$q_{y^{\star},t+1}^{\star} = 1 - (\mathbf{y}_{t+1}^{\star})^{\top} \boldsymbol{\Sigma}_{y^{\star}y^{\star}t}^{-1} \boldsymbol{\alpha}_{yt} [\mathbf{E}_{t}(\mathbf{r}_{t+1}) - r_{ft} \mathbf{1}] , \qquad (25)$$

where $\Sigma_{y^{\star}y^{\star}t}$ is the covariance matrix of the hedging-portfolio cash-flows. It is straightforward to verify that

$$\Sigma_{y^{\star}y^{\star}t} = \Sigma_{yrt} \Sigma_{rrt}^{-1} \Sigma_{ryt}$$
⁽²⁶⁾

holds.9

Hence,

$$q_{y^{\star},t+1}^{\star} = 1 - (\mathbf{y}_{t+1}^{\star})^{\top} (\boldsymbol{\Sigma}_{yrt} \boldsymbol{\Sigma}_{rrt}^{-1} \boldsymbol{\Sigma}_{ryt})^{-1} \boldsymbol{\Sigma}_{yrt} \boldsymbol{\Sigma}_{rrt}^{-1} [\mathbf{E}_t(\mathbf{r}_{t+1}) - r_{ft} \mathbf{1}] .$$
(27)

 $^{9}\mathrm{I}~\mathrm{have}$

$$\boldsymbol{\Sigma}_{y^{?}y^{?}t} = \mathrm{E}_{t}[\boldsymbol{\alpha}_{yt}(\mathbf{r}_{t+1} - \mathrm{E}_{t}(\mathbf{r}_{t+1}))(\mathbf{r}_{t+1} - \mathrm{E}_{t}(\mathbf{r}_{t+1}))^{\top}\boldsymbol{\alpha}_{yt}^{\top}] = \boldsymbol{\Sigma}_{yrt}\boldsymbol{\Sigma}_{rrt}^{-1}\boldsymbol{\Sigma}_{rrt}\boldsymbol{\Sigma}_{rrt}\boldsymbol{\Sigma}_{rrt}^{-1}\boldsymbol{\Sigma}_{ryt} = \boldsymbol{\Sigma}_{yrt}\boldsymbol{\Sigma}_{rrt}^{-1}\boldsymbol{\Sigma}_{ryt} \ .$$

Under the null of the multi-beta model, $E_t(\mathbf{r}_{t+1}) - r_{ft}\mathbf{1} = \boldsymbol{\Sigma}_{ryt}\boldsymbol{\lambda}_t$, and the expression above simplifies to

$$q_{y^{\star},t+1}^{\star} = 1 - (\mathbf{y}_{t+1}^{\star})^{\top} \boldsymbol{\lambda}_t .$$
⁽²⁸⁾

In other words, under the null, the projection of $q_{y,t+1}$ onto the augmented span of asset returns equals the minimum-variance kernel constructed using the hedging portfolio cashflows.

F. Hedging Portfolios and Tests of Asset-Pricing Models

One appealing property of the kernel $q_{y^*,t+1}^*$ is that, under the null, it is the minimumvariance admissible kernel. In other words, any other admissible kernel is at least as volatile as $q_{y^*,t+1}^*$. This minimum-volatility property is obviously appealing. When it comes to tests of overidentifying restrictions (the standard J test of Hansen, 1982), low volatility of the kernel implies low volatility of the pricing errors, $q_{y^*,t+1}^*\mathbf{r}_{t+1} - r_{ft}\mathbf{1}$. This, in turn, reduces the likelihood that a poor model is not rejected simply because the volatility of the candidate pricing kernel is high.

One additional feature that makes the kernel $q_{y^{\star},t+1}^{\star}$ appealing is that the statistics associated with tests of the Hansen-Jagannathan variance bounds (HJV) and the Hansen-Jagannathan distance (HJD) are the same, and can be interpreted in terms of comparisons of expected squared conditional Sharpe ratios of the unrestricted tangency portfolio and a restricted tangency portfolio.

The unconditional variance of $q_{y^*,t+1}^*$ equals the expectation of the squared conditional Sharpe ratio of the tangency portfolio obtained using the hedging portfolios, instead of all the assets available. Hence, when I test whether $q_{y^*,t+1}^*$ satisfies the HJ variance bound, I test whether $\operatorname{Var}(q_{t+1}^*) - \operatorname{Var}(q_{y^*,t+1}^*) > 0$. This is equivalent to a test that $\operatorname{E}(\operatorname{S}^2_{\tau t}) - \operatorname{E}(\operatorname{S}^2_{y^*\tau t}) > 0$ where $S_{y^*\tau t}$ is the Sharpe ratio of the restricted tangency portfolio.

The HJD test is a test that the projection of a candidate kernel onto the augmented span of asset returns and the minimum-variance kernel are sufficiently "close": under the null, the second moment of the difference between the minimum-variance kernel and the projection of the candidate kernel onto the augmented span of asset returns should be "small". In particular, using $q_{y^{\star},t+1}$ as the candidate kernel (note that the projection of $q_{y^{\star},t+1}^{\star}$ onto a constant and \mathbf{r}_{t+1} is $q_{y^{\star},t+1}^{\star}$ itself) the HJD statistic is the sample counterpart of

$$\mathbf{E}[(q_{t+1}^{\star} - q_{y^{\star},t+1}^{\star})^2] = \mathbf{E}\{\mathbf{E}_t(q_{t+1}^{\star} - q_{y^{\star},t+1}^{\star})^2\} = \mathbf{E}[\operatorname{Var}_t(q_{t+1}^{\star} - q_{y^{\star},t+1}^{\star})] .$$
(29)

Now note that

$$Cov_t(q_{t+1}^{\star} - q_{y^{\star},t+1}^{\star}) = [\mathbf{E}_t(\mathbf{r}_{t+1} - r_{ft}\mathbf{1})]^{\top} \boldsymbol{\Sigma}_{rrt}^{-1} \boldsymbol{\Sigma}_{rr} \boldsymbol{\alpha}_y^{\top} \boldsymbol{\Sigma}_{rr}^{-1} \boldsymbol{\alpha}_y [\mathbf{E}_t(\mathbf{r}_{t+1}) - r_{ft}\mathbf{1}]$$

$$= [\mathbf{E}_t(\mathbf{r}_{t+1}) - r_{ft}\mathbf{1}]^{\top} \boldsymbol{\alpha}_y^{\top} \boldsymbol{\Sigma}_{rr}^{-1} \boldsymbol{\alpha}_y [\mathbf{E}_t(\mathbf{r}_{t+1}) - r_{ft}\mathbf{1}]$$

$$= \operatorname{Var}_t(q_{y^{\star},t+1}^{\star}).$$
(30)

Hence

$$E[\operatorname{Var}_{t}(q_{t+1}^{\star} - q_{y^{\star},t+1}^{\star})] = E[\operatorname{Var}_{t}(q_{t+1}^{\star})] - E[\operatorname{Var}_{t}(q_{y^{\star},t+1}^{\star})]$$
$$= \operatorname{Var}(q_{t+1}^{\star}) - \operatorname{Var}(q_{y^{\star},t+1}^{\star}).$$
(31)

Up to this point, I have assumed the existence of a pricing kernel q_{t+1} which prices exactly the securities under consideration. I now investigate theories which postulate the form of q_{t+1} . For concreteness, I consider here four models: the International Static CAPM (IS– CAPM); the International CAPM in presence of deviations from PPP (I-CAPM (PPP)); the International CAPM in presence of currency risk (I-CAPM (SPOT)); and the International Intertemporal CAPM (II–CAPM). The IS–CAPM was first derived by Solnik (1974) and postulates a linear relationship between the cross-section of international expected excess equity returns and the excess return on a world market portfolio. The I-CAPM, as introduced by Adler and Dumas (1983), links nominal excess returns on international equity denominated in a reference currency to a world market portfolio and portfolios hedging against deviations from PPP. The II-CAPM is a combination of Merton's (1973) intertemporal CAPM and Adler and Dumas's (1983) international CAPM. Namely, the cross-section of nominal excess returns denominated in a reference currency is explained by three hedging funds: a nationless (or logarithmic) world market portfolio; portfolios hedging against deviations from PPP; and portfolios hedging against variations in the investment opportunity set of an international investor. Consider the II-CAPM in its discrete-time approximate version. The vector \mathbf{w}_t^l of the *l*-country (l = 1, ..., L + 1) optimal weights on the risky assets is given by

$$\mathbf{w}_{t}^{l} = \alpha^{l} \boldsymbol{\Sigma}_{rrt}^{-1} [\mathbf{E}_{t}(\mathbf{r}_{t+1}) - r_{ft} \mathbf{1}] + (1 - \alpha^{l}) \boldsymbol{\Sigma}_{rrt}^{-1} \mathbf{s}_{r\pi t}^{l} + \beta^{l} \boldsymbol{\Sigma}_{rrt}^{-1} \mathbf{s}_{ry_{k}t}^{l} , \qquad (32)$$

where $\mathbf{s}_{ry_k t}^l$ and $\mathbf{s}_{r\pi t}^l$ are the vectors of time-varying covariances between each country's security returns and the k-th state variable and level of inflation, respectively. Moreover, I define $\alpha^l \equiv -\frac{J_{Wt}^l}{J_{WWt}^l W_t^l}$ and $\beta^l \equiv (\alpha^l)^2 \times \frac{J_{Wyk}^l}{W_t^l}$.¹⁰ Let $\mathbf{w}_{\tau t}$ denote the $N \times 1$ vector of unscaled weights of the tangency portfolio

$$\mathbf{w}_{\tau t} = \boldsymbol{\Sigma}_{rrt}^{-1} [\mathbf{E}_t(\mathbf{r}_{t+1}) - r_{ft} \mathbf{1}] .$$
(33)

Let \mathbf{w}_{yt}^{l} and $\mathbf{w}_{\pi t}^{l}$ denote the $N \times 1$ vectors of unscaled weights hedging against variations in the investment opportunity set and against deviations from PPP, respectively, so that

$$\mathbf{w}_{yt}^{l} = \boldsymbol{\Sigma}_{rrt}^{-1} \mathbf{s}_{rykt}^{l} \quad \text{and}$$
(34)

$$\mathbf{w}_{\pi t}^{l} = \boldsymbol{\Sigma}_{rrt}^{-1} \mathbf{s}_{ryt}^{l} .$$
(35)

Then I can rewrite (32) as

$$\mathbf{w}_t^l = \alpha^l \mathbf{w}_{\tau t} + (1 - \alpha^l) \mathbf{w}_{\pi t}^l + \beta^l \mathbf{w}_{yt}^l .$$
(36)

Aggregating over countries and defining $\alpha^m = \frac{\sum_{l=1}^{L+1} \alpha^l W^l}{\sum_{l=1}^{L+1} W^l}, \ \alpha^l_{\pi} = \frac{(1-\alpha^l)W^l}{\sum_{l=1}^{L+1} W^l}, \ \alpha^l_y = \frac{\beta^l W^l}{\sum_{l=1}^{L+1} W^l}, \ I$ obtain

$$\mathbf{w}_{mt} = \alpha^m \mathbf{w}_{\tau t} + \sum_{l=1}^{L+1} \alpha_\pi^l \mathbf{w}_{\pi t}^l + \sum_{l=1}^{L+1} \alpha_y^l \mathbf{w}_{yt}^l .$$
(37)

Hence the tangency portfolio, which prices all security returns, is a combination of the global market portfolio and the portfolios hedging against deviations from PPP and movements in the investment opportunity set of an international investor. Notice that the II-CAPM collapses to the IS-CAPM when the investment opportunity set is constant and there are no deviations from PPP.¹¹ The II-CAPM collapses to the I-CAPM when the investment

¹⁰See Appendix B for a derivation of (32).

¹¹Note that in this case, using either nominal or real stock returns to test the CAPM returns the same results.

opportunity set is constant or, equivalently, when the weights associated with the hedging demands are set equal to zero. Moreover, when the inflation rate of country l, expressed in its home currency, is zero or non stochastic, the L + 1 inflation hedging funds of Adler and Dumas model collapse to L exchange rate hedging funds:¹² the I-CAPM (PPP) collapses to the I-CAPM (SPOT). The previous setup provides a rationale for using nominal returns in the analysis. The absence of money illusion also makes it possible to express nominal returns in a reference currency and, without loss of generality, in excess of a measurement currency risk-free rate.¹³ The problem of testing these different versions of the international CAPM simply reduces to the problem of testing that the tangency portfolio is mean-variance efficient. The international asset pricing models discussed above can be conveniently stated in terms of their assumptions on the nominal (normalized) minimum-variance kernel q^* .¹⁴ Hence, I define the vector \mathbf{y}_{t+1} as

$$\mathbf{y}_{t+1} \equiv [y_{m,t+1}, \mathbf{y}_{f,t+1}, \mathbf{y}_{\pi,t+1}, \mathbf{y}_{h,t+1}]^{\top},$$

where $y_{m,t+1}$ is the rate of return on the world market portfolio, $\mathbf{y}_{f,t+1}$ is the $L \times 1$ vector of logarithmic changes of the rates of appreciation of the measurement currency, $\mathbf{y}_{\pi,t+1}$ is the $(L+1) \times 1$ vector of innovations in the inflation rates, and $\mathbf{y}_{h,t+1}$ is the $K \times 1$ vector of demands hedging against variations in the investment opportunity set, where K represents the number of economic factors used in the analysis. The corresponding mimicking-portfolio returns, \mathbf{y}_{t+1}^{\star} , are formed using the methodology described in Section I.D.

Consider, for example, the II-CAPM. The corresponding normalized minimum-variance

¹³In the IS-CAPM, translating returns into a new currency and measuring excess returns relative to the new currency risk-free rate would leave the intercept term equal to zero. In the I-CAPM and II-CAPM, the new currency foreign exchange premium would be replaced by the old currency exchange risk premium. Nonetheless, the introduction of conditioning information and the expansion of the set of primitive securities to include managed portfolios might be affected by the choice of the measurement currency. Hence, the pricing implications delivered by alternative asset pricing models might differ according to the reference currency considered. Dumas and Solnik (1995) found that the choice of the measurement currency did not affect their conclusions in terms of rejection and acceptance of the international CAPM.

¹⁴See Dumas and Solnik (1995) for a similar interpretation of international asset pricing models.

¹²See, for example, Solnik (1974), Sercu (1980), and Grauer, Litzenberger and Stehle (1976).

pricing kernel, q_{t+1}^{\star} , can be written as

$$q_{t+1}^{\star} = 1 - y_{m,t+1}^{\star} \lambda_{mt} - \mathbf{y}_{\pi,t+1}^{\star\top} \boldsymbol{\lambda}_{\pi t} - \mathbf{y}_{h,t+1}^{\star\top} \boldsymbol{\lambda}_{ht} , \qquad (38)$$

or, when inflation rates in each country are non-random, as

$$q_{t+1}^{\star} = 1 - y_{m,t+1}^{\star} \lambda_{mt} - \mathbf{y}_{f,t+1}^{\star\top} \boldsymbol{\lambda}_{ft} - \mathbf{y}_{h,t+1}^{\star\top} \boldsymbol{\lambda}_{ht} , \qquad (39)$$

where λ_{mt} , λ_{ft} , $\lambda_{\pi t}$ and λ_{ht} are commensurable coefficient vectors.

Using (5), I have

$$\mathbf{E}_{t}(\mathbf{r}_{t+1}) - \mathbf{1}\mathbf{r}_{ft} = \mathbf{E}_{t}(\mathbf{r}_{t+1}y_{m,t+1}^{\star})\lambda_{mt} + \mathbf{E}_{t}(\mathbf{r}_{t+1}\mathbf{y}_{m,t+1}^{\star\top})\boldsymbol{\lambda}_{\pi t} + \mathbf{E}_{t}(\mathbf{r}_{t+1}\mathbf{y}_{h,t+1}^{\star\top})\boldsymbol{\lambda}_{ht}$$
(40)

and

$$\mathbf{E}_{t}(\mathbf{r}_{t+1}) - \mathbf{1}\mathbf{r}_{ft} = \mathbf{E}_{t}(\mathbf{r}_{t+1}y_{m,t+1}^{\star})\lambda_{mt} + \mathbf{E}_{t}(\mathbf{r}_{t+1}y_{f,t+1}^{\star\top})\boldsymbol{\lambda}_{ft} + \mathbf{E}_{t}(\mathbf{r}_{t+1}y_{h,t+1}^{\star\top})\boldsymbol{\lambda}_{ht} .$$
(41)

Let $\boldsymbol{\beta}_{mt} \equiv \mathbf{E}_t(\mathbf{r}_{t+1}\boldsymbol{y}_{m,t+1}^{\star}), \boldsymbol{\beta}_{ft} \equiv \mathbf{E}_t(\mathbf{r}_{t+1}\mathbf{y}_{f,t+1}^{\star\top}), \boldsymbol{\beta}_{\pi t} \equiv \mathbf{E}_t(\mathbf{r}_{t+1}\mathbf{y}_{\pi,t+1}^{\star\top}), \text{ and } \boldsymbol{\beta}_{ht} \equiv \mathbf{E}_t(\mathbf{r}_{t+1}\mathbf{y}_{h,t+1}^{\star\top})$ denote the (arrays of) the conditional "betas" associated with the economic variables \mathbf{y} . I can rewrite (40) and (41) to obtain the international intertemporal CAPM linear pricing results

$$\mathbf{E}_{t}(\mathbf{r}_{t+1}) - \mathbf{1}\mathbf{r}_{\mathsf{ft}} = \boldsymbol{\beta}_{\mathsf{mt}}\lambda_{\mathsf{mt}} + \boldsymbol{\beta}_{\mathsf{\pi t}}\boldsymbol{\lambda}_{\mathsf{\pi t}} + \boldsymbol{\beta}_{\mathsf{ht}}\boldsymbol{\lambda}_{\mathsf{ht}}$$
(42)

and

$$E_t(\mathbf{r}_{t+1}) - \mathbf{1}\mathbf{r}_{ft} = \boldsymbol{\beta}_{mt}\lambda_{mt} + \boldsymbol{\beta}_{ft}\boldsymbol{\lambda}_{ft} + \boldsymbol{\beta}_{ht}\boldsymbol{\lambda}_{ht} .$$
(43)

Equations (42) and (43) state that the conditional risk premium on any asset is a linear combination of the conditional risk premia on the different sources of economic and financial risks. The previous equations collapse to the IS-CAPM when PPP holds and hedging demands are equal to zero while they reduce to the Adler and Dumas I-CAPM when hedging demands are equal to zero.¹⁵

¹⁵Specifically, in testing the II-CAPM, I overparameterize the model and use demands hedging against inflation and foreign exchange risk simultaneously.

II. Estimation

This section describes the methodology I use in the empirical analysis. First, I illustrate the approach taken to document time variation in the first and second moments of international asset returns. Second, I illustrate the estimation of the coefficients of the minimum-variance kernel. Third, I illustrate the estimation of the economic risk premia using two approaches based on the minimum-variance kernel and mimicking portfolios. Fourth, I illustrate how the explicit asset pricing models discussed in the previous section are tested.

A. Selecting Instruments

Let \mathbf{z}_t denote a $J \times 1$ vector of instruments whose realizations belong to the information set at the beginning of each investment period. Without loss of generality, I assume the first element of \mathbf{z}_t to be unity, $z_{1t} = 1$. I model the conditional mean and conditional volatility of asset returns as linear functions of the instruments. Namely, I assume

$$E_t(\mathbf{r}_{t+1}) = \mu_{r1} + \mu_{r2} z_{2t} + \dots + \mu_{J1} z_{Jt} , \text{ and}$$
$$E_t[|\mathbf{r}_{t+1} - E_t(\mathbf{r}_{t+1})|] = v_{r1} + v_{r2} z_{2t} + \dots + v_{J1} z_{Jt} .$$
(44)

Hence, the mean and volatility coefficients can be estimated by exactly-identified GMM.

B. The Minimum-Variance Kernel

I postulate that the coefficients α_{0t} and $\boldsymbol{\alpha}_t$ of the minimum-variance pricing kernel are linear functions of the instruments \mathbf{z}_t ;

$$\alpha_{0t} = 1 - \mathbf{E}_t(\mathbf{r}_{t+1})^\top \boldsymbol{\alpha}_t = \alpha_0(\mathbf{z}_t) , \qquad (45)$$

and

$$\boldsymbol{\alpha}_{t} = -\boldsymbol{\Sigma}_{rrt}^{-1}[\mathbf{E}_{t}(\mathbf{r}_{t+1}) - r_{ft}\mathbf{1}] = \boldsymbol{\alpha}(\mathbf{z}_{t}) , \qquad (46)$$

where $\boldsymbol{\alpha}_j$, $j = 1, \ldots, J$, are $N \times 1$ coefficient vectors.

Note that this approach can be interpreted as the scaling of asset returns with instruments (or the expansion of the set of securities to include managed portfolios) which is common in the asset pricing literature.¹⁶ In fact, I can write

$$\mathbf{E}_t(q_{t+1}^\star)\mathbf{z}_t = \mathbf{z}_t , \text{ and}$$
(47)

$$\mathbf{E}_t(q_{t+1}^{\star}\mathbf{r}_{t+1}) \otimes \mathbf{z}_t = r_{ft}\mathbf{1} \otimes \mathbf{z}_t .$$
(48)

Assuming stationarity, and applying the law of iterated expectations, I have

$$\mathbf{E}(q_{t+1}^{\star}\mathbf{z}_t) = \mathbf{E}(\mathbf{z}_t), \text{ and}$$
(49)

$$\mathbf{E}(q_{t+1}^{\star}\mathbf{r}_{t+1}\otimes\mathbf{z}_{t}) = \mathbf{E}(r_{ft}\mathbf{1}\otimes\mathbf{z}_{t}) .$$
(50)

The two conditions above ensure that, conditioning on \mathbf{z}_t , q_{t+1}^{\star} has mean one and correctly prices the securities under consideration.

It is convenient at this point to define

$$\boldsymbol{\iota}_t \equiv \begin{bmatrix} 1\\ r_{ft} \mathbf{1} \end{bmatrix}$$
(51)

and let $\boldsymbol{\iota}_t^z \equiv \boldsymbol{\iota}_t \otimes \mathbf{z}_t$. Also, I define

$$\mathbf{r}_{a,t+1} \equiv \begin{bmatrix} 1\\ \mathbf{r}_{t+1} \end{bmatrix}$$
(52)

and $\mathbf{r}_{a,t+1}^{z} \equiv \mathbf{r}_{a,t+1} \otimes \mathbf{z}_{t}$. Hence, I can rewrite (49) and (50) as

$$\mathbf{E}(q_{t+1}^{\star}\mathbf{r}_{a,t+1}^{z}) = \mathbf{E}(\boldsymbol{\iota}_{t}^{z}) .$$
(53)

The minimum-variance kernel satisfying (53) has the form $q_{t+1}^{\star} \equiv \mathbf{r}_{a,t+1}^{z^{\top}} \boldsymbol{\alpha}_{a}^{z}$, where $\boldsymbol{\alpha}_{a}^{z}$ is an (N+1)J coefficient vector. I have

$$\boldsymbol{\alpha}_{a}^{z} = [\mathrm{E}(\mathbf{r}_{a,t+1}^{z}\mathbf{r}_{a,t+1}^{z\top})]^{-1}\mathrm{E}(\boldsymbol{\iota}_{t}^{z}) .$$
(54)

¹⁶This scaling procedure has an intuitive interpretation. The scaled returns are the returns on managed portfolios in which the manager invests more or less according to the signals z_t .

Hence, the minimum-variance kernel satisfying (53) has the form

$$q_{t+1}^{\star} = (\alpha_{a1}^{z} + \alpha_{a2}^{z} z_{2t} + \ldots + \alpha_{aJ}^{z} z_{Jt}) + (\alpha_{a,J+1}^{z} + \alpha_{a,J+2}^{z} z_{2t} + \ldots + \alpha_{a,2J}^{z} z_{Jt})r_{1,t+1} + \ldots + (\alpha_{a,NJ+1}^{z} + \alpha_{a,NJ+2}^{z} z_{2t} + \ldots + \alpha_{a,J(N+1)}^{z} z_{Jt})r_{N,t+1}, \qquad (55)$$

which is equivalent to imposing the assumptions (45) and (46). The analysis above still applies when the positivity constraint is imposed.¹⁷ In this case the minimum-variance kernel has the form $\tilde{q}_{t+1} \equiv (\mathbf{r}_{a,t+1}^{z^{\top}} \tilde{\boldsymbol{\alpha}}_{a}^{z})^{+}$.

Note that when the set of instruments used in the analysis only contains a constant, this approach can be interpreted as projecting the minimum–variance kernel on the set of primitive securities.

C. Economic Risk Premia

In order to estimate the conditional risk premia associated with the variables y_{kt} , I use two approaches. First, I consider the conditional covariance between the minimum-variance kernel and y_k . Second, I construct mimicking portfolios and I estimate their conditional risk premia.

In implementing the first approach, I assume

$$\lambda_{kt}^{\star} \equiv -\operatorname{Cov}_t(q_{t+1}^{\star}, y_{k,t+1}) = -\operatorname{E}_t[(q_{t+1}^{\star} - 1)y_{k,t+1}] = \lambda_{k1} + \lambda_{k2}z_{2t} + \ldots + \lambda_{kJ}z_{Jt} .$$
(56)

Without loss of generality, I also assume $\operatorname{Var}(z_{jt}) = 1$. Hence the coefficients λ_{kj} can be interpreted as the change in the conditional risk premium for a one-standard-deviation change in the instrument. The assumption that the conditional risk premia are determined by the set of instruments \mathbf{z}_t is quite natural: the conditional risk premia assigned by the minimumvariance kernel are the mean cash flows generated by the hedging portfolio financed at the

¹⁷Appendix C shows how to estimate economic risk premia under the no arbitrage condition of nonnegativity of the normalized pricing kernel.

riskless rate. Hence, if the variables in \mathbf{z}_t are predictors of asset returns, they should also predict excess returns on the hedging portfolios. In fact, modeling the time-variation of risk premia in this fashion is common to other studies (Ferson and Harvey (1991), for example). Note that, without loss of generality, I can assume $\mathbf{E}(z_{jt}) = 0, i = 2, ..., J$. This means that

$$\mathbf{E}(\lambda_{kt}^{\star}) = \lambda_{k1} \ . \tag{57}$$

The distinction between conditional and unconditional risk premia is important because, even if the unconditional premium is close to zero, the conditional premia may take values over time which are significantly different from zero.

When the positivity restriction is imposed, I have

$$\hat{\lambda}_{kt} \equiv -\text{Cov}_t(\tilde{q}_{t+1}, y_{k,t+1}) = -\text{E}_t[(\tilde{q}_{t+1} - 1)y_{k,t+1}] = \lambda_{k1} + \lambda_{k2}z_{2t} + \ldots + \lambda_{kJ}z_{Jt} .$$
(58)

The second approach to the estimation of the economic risk premia is based on the construction of hedging portfolios. I postulate that the coefficients α_{0y_kt} and α_{y_kt} of the hedging portfolio are linear functions of the instruments \mathbf{z}_t . Hence, I have $y_{k,t+1}^{\star} = \boldsymbol{\alpha}_{ay_k}^z \mathbf{r}_{a,t+1}^z$, where

$$\boldsymbol{\alpha}_{ay_k}^z = \left[\mathbf{E}(\mathbf{r}_{a,t+1}^z \mathbf{r}_{a,t+1}^{z\top}) \right]^{-1} \mathbf{E}(\mathbf{r}_{a,t+1}^z y_{k,t+1}) \ . \tag{59}$$

As with the coefficients of the minimum-variance kernel, one can show that this approach is equivalent to projecting the risk variables on both the returns on the primitive securities and the cash flows of dynamic strategies. In particular, when $\mathbf{z}_t = z_{1t} = 1$ this approach is equivalent to projecting the risk variables only on the set of primitive securities.

Finally, I estimate the Sharpe ratios of the hedging portfolios, since they might differ substantially from the conditional risk premia (see discussion above). Hence, I take the ratio between the risk premium and the volatility of the hedging portfolios cash flows.

D. Tests of International Asset-Pricing Models

Each of the international asset-pricing models I considered has specific pricing implications, as previously discussed. For convenience, I discuss my tests with respect to the implications of the static international capital asset pricing model (IS-CAPM). The approach to test the other models is analogous.

Recall that the IS-CAPM implies that the portfolio hedging global market risk prices all securities. Using a standard GMM test I test whether a pricing kernel linear in the cash flows of this portfolio, $q_{m,t+1}^{\star}$, prices all securities. I have $q_{m,t+1}^{\star} = \mathbf{y}_{ma,t+1}^{\star} \boldsymbol{\alpha}_{ma}^{z}$, where

$$\boldsymbol{\alpha}_{ma}^{z} = \mathrm{E}(\mathbf{y}_{ma,t+1}^{\star}\mathbf{y}_{ma,t+1}^{\star\top})^{-1}\mathrm{E}(\boldsymbol{\iota}_{mt})$$
(60)

and

$$\mathbf{y}_{ma,t+1}^{\star} \equiv \begin{bmatrix} \mathbf{z}_t \\ \mathbf{r}_{a,t+1}^{z^{\top}} \boldsymbol{\alpha}_{ay_m}^z \end{bmatrix}$$
(61)

$$\boldsymbol{\iota}_{mt} \equiv \begin{bmatrix} \mathbf{z}_t \\ \boldsymbol{\iota}_t^{z^\top} \boldsymbol{\alpha}_{ay_m}^z \end{bmatrix}.$$
(62)

I test the moment conditions

$$\mathbf{E}(q_{m,t+1}^{\star}\mathbf{r}_{a,t+1}^{z}) = \mathbf{E}(\boldsymbol{\iota}_{t+1}^{z}) .$$
(63)

This corresponds to the test of overidentifying restrictions pioneered by Hansen and Singleton (1982). I have J+K coefficients in the vector $\boldsymbol{\alpha}_{ma}^{z}$ and J+NJ moment conditions, for a total of NJ - K overidentifying restrictions. If the test failed, this would mean that the pricing errors generated by the IS-CAPM are statistically significant, and the model is rejected.

Second, I compare the standard deviation of $q_{m,t+1}^{\star}$ to the standard deviation of the minimum-variance normalized kernel, q_{t+1}^{\star} . Namely, I estimate the difference

$$\sqrt{\operatorname{Var}(q_{m,t+1}^{\star})} - \sqrt{\operatorname{Var}(q_{t+1}^{\star})} .$$
(64)

This corresponds to the HJV test. Since the variances of the two kernels correspond to the average squared Sharpe ratios of the global market-hedging portfolio and the tangency portfolio, this test is equivalent to a test of the mean-variance efficiency of the world markethedging portfolio.

Third, I calculate the standard deviation of the difference between $q_{m,t+1}^{\star}$ and q_{t+1}^{\star} ,

$$\sqrt{\operatorname{Var}(q_{t+1}^{\star} - q_{m,t+1}^{\star})}$$
 (65)

This corresponds to the HJD test.¹⁸ If the test failed, this would mean that the difference between the pricing kernel generated by the IS-CAPM and any admissible kernel is statistically significant, and the model is rejected.

III. Data

This section illustrates the data used in the empirical analysis. The period considered is April 1970 through October 1998 for stock returns and economic variables and March 1970 through September 1998 for instrumental variables. Data are monthly. The starting and ending dates for the sample are dictated by macroeconomic and financial data availability.

A. Asset Returns

I use the Morgan Stanley Capital International (MSCI) national equity indices. The nominal returns are denominated in U.S. dollars and are calculated with dividends. All indices have a common basis of 100 in December 1969. The indices are constructed using the Laspeyres method, which approximates value weighting.¹⁹ U.S. dollar returns are calculated by using the closing European interbank currency rates from MSCI. I choose the four countries with the largest market capitalization: United States; United Kingdom; Japan; and Germany.²⁰ Table I shows summary statistics for monthly returns on stock indices from MSCI.

¹⁸Since I allow a constant in $q_{m,t+1}^{?}$, $E(q_{m,t+1}^{?}) = E(q_{t+1}^{?}) = 1$. Hence, $\sqrt{E(q_{t+1}^{?} - q_{m,t+1}^{?})^2}$, which is the Hansen-Jagannathan distance, equals $\sqrt{Var(q_{t+1}^{?} - q_{m,t+1}^{?})}$.

¹⁹See MSCI Methodology & Index Policy for a detailed description of MSCI's indices and properties.

 $^{^{20}\}mathrm{As}$ of 1996, the market capitalization weight for these countries is 76.2% of the market capitalization world.

B. Economic Variables and Instruments

I use inflation rates, spot exchange rates, and the rate of return on the world market portfolio as the relevant sources of risk in this study. Consumer price indices are from International Financial Statistics (IFS) and are denominated in local currency. Spot exchange rates are from MSCI. The world equity market index is a value–weighted combination of the country returns tracked by MSCI. I proxy foreign exchange risk with logarithmic changes in the spot exchange rates (SPOT); inflation risk with an ARIMA(0,1,1) for inflation (INFL); and global market risk with the level of world equity returns (WLDMK). Note that the variable INFL represents unexpected inflation and that the variable SPOT represents innovations in the exchange rate if I assume that spot rates follow a random walk. Table II contains summary statistics for the relevant risk variables. In choosing the set of instruments, I concentrate on a set of variables which have been previously used in tests of multiple-beta models and/or in studies of stock-return predictability.²¹ These variables are statistically significant in multivariate predictive regressions of means and volatilities and/or they have special economic significance. The instruments include a constant, a January dummy, and the following five variables:

DINFLUS is the lagged difference in the U.S. monthly rate of inflation (IFS).

EURO represents the one-month Eurodollar deposit rate (DRI) and performs as the conditionally nominal risk-free asset in the analysis.

USDIVYLD denotes the U.S. monthly dividend yield (MSCI) in excess of the 1-month Eurodollar deposit rate. Specifically, the monthly dividend yield is equal to 1/12 of the ratio between the previous year dividend and the index at the end of each month.

WLDMK denotes the lagged value of the world stock market monthly returns (MSCI).

DEFPREM denotes the U.S. default premium as given by the return difference between Moody's Baa-rated and Aaa-rated bonds (SBBI Yearbook).

²¹See, for example, Ferson and Harvey (1993), Dumas and Solnik (1995), and De Santis and Gérard (1998).

I select as instruments the previous variables as a proxy for the information investors use to set prices in the market.

Table III presents summary statistics of the instrumental variables.

IV. Selecting Instruments

In this section I select a set of instruments by performing an analysis of predictability. Specifically, I look at the ability of some variables used in previous studies of international asset pricing to predict variation in the first and second conditional moments. This preliminary analysis is relevant for three reasons. First, it identifies the information set of an international investor. Second, it makes it possible to study the patterns of time variation in the conditional risk premia. Third, it provides a rationale for including a third hedging portfolio in alternative international CAPM specifications besides portfolios hedging against global market and inflation (or currency) risks.

Table IV presents my results. I report the coefficients of the mean and variance equations, as well as three statistics: $\bar{\mu}_r$ is the average slope coefficient in the mean equations; \bar{v}_r is the average slope coefficient in the variance equations; and $\bar{\mu}_r - \bar{v}_r$ is the difference between the two average slope coefficients. These statistics provide an indication of the net effect of the instruments on the investment opportunity set.

The following patterns emerge from this analysis (see especially Panel C):

The lagged change in the U.S. inflation rate (DINFLUS) has a negative and significant average impact on returns and a negative, but insignificant, average impact on return volatility. The net effect on the investment-opportunity set is strongly negative.

The Eurodollar deposit rate (EURO) has a positive and significant average impact on returns and a negative and significant average impact on return volatility. The net effect is strongly positive.

The U.S. dividend yield in excess of the EURO (USDIVYLD) has a positive and

significant average effect on returns and a negative and significant overall effect on volatility. The net effect is positive.

The lagged world market equity return (WLDMK) has an overall positive, but insignificant, impact on returns. The impact on volatility is negative and partially significant. The net effect is positive but not large.

The U.S. default premium (DEFPREM) positively affects returns and is partially significant. The overall effect on volatility is negative and partially significant. The net effect is positive.

In summary, I can rank the net effects of the different variables on the investmentopportunity set as follows (from largest to smallest): USDIVYLD, EURO, DEFPREM, WLDMK, DINFLUS.²²

V. Risk Premia and Sharpe Ratios

In this section I report estimates of the risk premia associated with the economic variables and look at their patterns of time variation.²³ I use the instrumental variables selected in the previous section to document the patterns of time variation of the conditional premia.

Table V reports estimates of the coefficients of the economic risk premia estimated using the minimum-variance kernel q_{t+1}^{\star} . Since the instruments are demeaned, the intercept term can be interpreted as the unconditional risk premium on $y_{k,t+1}$. I do not report coefficient estimates of the economic risk premia estimated using the non-negative minimum-variance kernel \tilde{q}_{t+1} because there is no substantial difference with respect to the estimates reported in Table VI. Table VII reports coefficient estimates of the economic risk premia estimated using the hedging portfolios.

²²Note that Germany, among all countries in the study, exhibits the weakest patterns of predictability.

²³Note that the instruments do not coincide with the past values of the economic variables included in the analysis, with the exception of the world market portfolio.

As noted in section I.D. above, the expected excess cash flows on the hedging portfolios coincide with risk premia assigned by q_{t+1}^* . Yet, the realized excess cash flows on the hedging portfolios in general differ from $(q_{t+1}^* - 1)y_{k,t+1}$. Hence, the estimates of the unconditional risk premia using the " q^* " and the "hedging portfolio" approaches will coincide, although their standard errors may differ. In addition, the impact of the conditioning variables on the conditional risk premia will also differ. Note that when y^* and q^* are mapped onto the space spanned by the primitive securities and are estimated inside of the algorithm, the standard errors of the unconditional risk premia coincide (see Panel A of Tables V and VI).

The tables report two sets of t-ratios. The first t-ratio is obtained using a two-step procedure: I first estimate the coefficients of q_{t+1}^{\star} and $y_{k,t+1}^{\star}$; I then estimate the risk premia by exactly-identified GMM algorithm. The second t-ratio is obtained estimating all parameters inside the GMM algorithm. The reason for the two separate approaches is that I am concerned with the large number of estimated parameters when the coefficients of the minimum-variance kernels and the hedging portfolios are estimated by GMM. As it turns out, the t-ratios change only marginally across the two procedures. In all tests, standard errors are adjusted for heteroskedasticity and serial correlation.

The results are also very similar across estimation methods. The main difference is that the estimation based on the q^* tends, in most of the cases, to yield tighter standard errors.

The following patterns emerge from the tables:

The unconditional inflation risk premia are negative (except for Germany) but not significant. When managed portfolios are ruled out from the analysis, only U.S. inflation seems to be priced and to be significant across estimation techniques. No significant pattern of time variation emerges from the analysis of the foreign inflation conditional premia.

The unconditional foreign exchange risk premia are negative. Even if I am not using any equilibrium model to compute economic risk premia, this result is consistent with the predictions of Adler's and Dumas' international CAPM. They show that if the degree of risk aversion of investors is greater than one, foreign exchange risk premia should be negative. Nonetheless, the unconditional estimates of currency risk premia are not statistically significant. This result also supports Adler and Dumas (1995) and De Santis and Gerard (1998). I also find time variation in the estimates of the conditional premia statistically significant.

The unconditional global market risk premium is positive and statistically significant across estimation techniques. This result seems to be robust and does not support the analysis of Adler and Dumas (1995) or De Santis and Gérard (1998). They actually find that the world market unconditional risk premium is positive but insignificant. On the other hand, my findings are supported by the work of Hodrick, Ng, and Sengmüller (1999). Moreover, I find significant time variation in the global market conditional risk premium.

In addition, I estimated the unconditional and conditional premia implied by the set of economic and financial factors. Estimation results (not reported in the paper) do not provide any evidence of statistical significance of these premia. Even if the factors exhibit some non-trivial patterns of predictability for the first and second moment of asset returns, they do not seem to be priced.

In summary, I find that the signs of the risk premia associated with foreign exchange and inflation risks are largely consistent with the theoretical predictions of several models of international asset pricing. At the same time, foreign inflation unconditional and conditional risk premia and foreign exchange unconditional risk premia are imprecisely estimated. On the contrary, I document significant patterns of time variation of foreign exchange premia and find that constant and time-varying global market risk premia are precisely estimated. The remaining economic and financial factors are not priced, both unconditionally and conditionally.

The risk premia estimated above coincide with the Sharpe ratios of exact mimicking portfolios. But, in general, economic factors can be tracked only imperfectly by asset returns. Hence, in order to obtain Sharpe ratios on traded portfolios I need to standardize the estimates obtained above by the volatility of the approximate mimicking portfolio returns. The composition of the mimicking portfolios is estimated separately from the Sharpe ratios.

Table VII presents the Sharpe ratios on the eight hedging portfolios. I find that the volatilities of the mimicking portfolios cash flows are strongly significant. On the other hand, given the statistical insignificance of the mean estimates, the Sharpe ratios on the hedging portfolios are insignificant, with the exception of the global market portfolio.

VI. Hedging Demands and Tests of International Asset Pricing Models

In this section I discuss the results of tests of four explicit asset pricing models: IS-CAPM, I-CAPM (PPP), I-CAPM (SPOT), and II-CAPM.²⁴ Results of the tests are presented in Table VIII and in Table IX. The tests are performed using the full set of instruments \mathbf{z}_t ("With conditioning information").

In Table VIII, I report the χ^2 statistic associated with a test of the overidentifying restrictions, the difference between the standard deviation of the candidate pricing kernel and the standard deviation of q_{t+1}^{\star} , the HJV statistic, and the standard deviation of the difference between the candidate pricing kernel and q_{t+1}^{\star} , the HJD statistic. I also report the p-values associated with the χ^2 test, and the t-ratios associated with the HJV and HJD statistics.

In the test of overidentifying restrictions, the coefficients of the candidate kernel are estimated by GMM, although the composition of the mimicking portfolios is estimated separately, outside of the GMM algorithm. In the other two tests, the coefficients of the candidate kernel are estimated separately.

I find that the χ^2 tests do not reject all five models conditionally. In particular, the ²⁴The additional mimicking portfolios considered in the II-CAPM specification are obtained via an OLS regression of DINFLUS, EURO, USDIVYLD and DEFPREM on the augmented span of managed portfolio returns. II-CAPM(PPP) can be weakly rejected at the 5% significance level but not at the 1% level.

On the contrary, tests of the HJ bounds and of the HJ distance measure indicate that all the models are rejected by the data – the standard deviation of the candidate kernels is always substantially lower than that of q_{t+1}^* . Nonetheless, as shown in Table IX, there are differences in performance. The IS-CAPM generates the least volatile pricing kernel. The I-CAPM (SPOT) of Grauer, Litzenberger, and Stehle (1976) generates pricing kernels with somewhat higher volatility. The second most volatile pricing kernel is generated by the Adler and Dumas model. The most volatile kernel is generated by the II-CAPM (PPP). The difference between the volatility of the II-CAPM (PPP) and I-CAPM (SPOT) kernels is substantial (68 percent) while the difference between the volatility of the II-CAPM (PPP) and I-CAPM (PPP) kernels is close to 16 percent. Note that the standard deviation of a pricing kernel coincides with the average Sharpe ratio of the tangency portfolio constructed using the underlying set of assets. Hence, the HJV statistic can be interpreted as the difference between two average Sharpe ratios.

Overall, the evidence from these tests is that while all five models are not rejected by the Hansen and Singleton (1982) χ^2 test of overidentifying restrictions, the same models are formally rejected by the tests of HJ variance bounds and HJ distance. These results are in sharp contrast with the findings of Dumas and Solnik (1995) as well as De Santis and Gérard (1998). These authors do not reject the international CAPM in its conditional version on the basis of tests of overidentifying restrictions and cross-equations restrictions. The use of test statistics that do not reward the variability of alternative admissible pricing kernels allows me to reject the conditional international CAPM in all its specifications.

I also investigate the size and the significance of the unconditional demands induced by hedging against global market, inflation, and foreign exchange risks. The estimates of these hedging demands correspond to the coefficients of the mimicking portfolios that appear in the kernel specification described in Section II.D. The hedging demands are normalized to sum up to one. Standard errors are computed using the delta method.

As shown in Table X, the scaled hedging demands are precisely estimated. In particular,

the coefficients associated with the cash-flows of the inflation hedging portfolios are large in magnitude and statistically significant. This result further explains why the international intertemporal CAPM in presence of deviations from PPP has better pricing implications than the other asset pricing models.²⁵

VII. Conclusions

This paper presents a new approach for the estimation of risk premia associated with observable sources of risk, which is based on the moments of the minimum-variance kernel of Hansen and Jagannathan (1991). I also provide extensive evidence on the performance of four explicit asset pricing models: the IS-CAPM; the I-CAPM (PPP); the I-CAPM (SPOT); and the II-CAPM in presence of deviations from PPP.

In sharp contrast with Dumas and Solnik (1995) and De Santis and Gérard (1998), but in line with Hodrick, Ng, and Sengmüller (1999), I find the global market risk is priced both conditionally and unconditionally.

All international asset-pricing models are formally rejected by the data when the testing methodology is stringent enough. In addition, the II-CAPM (PPP) that I construct and test using mimicking portfolios outperforms the IS-CAPM. It generates a pricing kernel which is 72% more volatile than the one of the IS-CAPM. For these models, the differences is Hansen & Jagannathan variance bounds and distance measures are large and statistically significant. Hence, I add to the existing literature showing that, introducing deviations from PPP and dynamic hedging, the II-CAPM (PPP) is able to generate more accurate pricing implications than the other versions of the international CAPM. My findings show that the II-CAPM (PPP) outperforms the competitive international asset-pricing models because most of the economic and financial factors considered significantly affect the first and second conditional moments of asset returns. Finally, the result that the international CAPM in its alternative

²⁵The inflation-mimicking portfolios make the II-CAPM(PPP) kernel more volatile and hence closer to the bounds of the minimum-variance kernel.

specifications does not hold is consistent with the evidence of at least mild segmentation of international equity markets and is supported by Hodrick, Ng, and Sengmüller (1999).

Future work should investigate the sensitivity of these results to the level of aggregation of asset returns, to the choice of the investment opportunity set of an investor, and to the measurement currency used in the analysis.

Appendix A

A. Unobservable Factors: Principal Components

One common approach to the study of asset returns is based on principal-component analysis. Let \mathbf{S}_t denote the matrix whose columns are the orthonormal eigenvectors of $\boldsymbol{\Sigma}_{rrt}$. Since $\mathbf{S}_t^{\top} \mathbf{S}_t = \mathbf{I}$, I have

$$\mathbf{r}_{t+1} = \mathbf{S}_t^{\top} \mathbf{S}_t \mathbf{r}_{t+1} \equiv \mathbf{S}_t^{\top} \mathbf{r}_{p,t+1} , \qquad (66)$$

where $\mathbf{r}_{p,t+1} \equiv \mathbf{S}_t \mathbf{r}_{t+1}$ is the vector of orthogonal factor-portfolio returns.²⁶ This makes it possible to rewrite the kernel q_{t+1}^{\star} as

$$q_{t+1}^{\star} = 1 - [\mathbf{r}_{t+1} - \mathbf{E}_t(\mathbf{r}_{t+1})]^{\top} \mathbf{S}_t^{\top} \mathbf{S}_t \mathbf{\Sigma}_{rrt}^{-1} \mathbf{S}_t^{\top} \mathbf{S}_t [\mathbf{E}_t(\mathbf{r}_{t+1}) - r_{ft} \mathbf{1}] = 1 - [\mathbf{r}_{p,t+1} - \mathbf{E}_t(\mathbf{r}_{p,t+1})]^{\top} \mathbf{V}_t^{-1} [\mathbf{E}_t(\mathbf{r}_{p,t+1}) - r_{ft} \mathbf{S}_t^{\top} \mathbf{1}],$$
(67)

where \mathbf{V}_t is a diagonal matrix whose v_{iit} element is the *i*-th eigenvalue of Σ_{rrt} .

Three main insights can be developed based on the expressions above. First, the minimumvariance kernel is a linear combination of the factor-portfolio returns

$$q_{t+1}^{\star} = 1 - \sum_{i=1}^{N} \frac{r_{pi,t+1} - \mathcal{E}_t(r_{pi,t+1})}{\sqrt{v_{iit}}} S_{pit} , \qquad (68)$$

where S_{pit} is the Sharpe ratio on the *i*-th factor-portfolio. Hence, the (standardized) risk premia associated with the factor portfolios are the coefficients relating the (standardized) innovations in the factor portfolio returns to the minimum-variance kernel. Moreover, equation (68) highlights how the variance of the minimum-variance kernel, and hence the squared Sharpe ratio of the tangency portfolio, can be decomposed according to the squared Sharpe ratios of the factor portfolios

$$\operatorname{Var}_{t}(q_{t+1}^{\star}) = S_{\tau t}^{2} = \sum_{i=1}^{N} S_{pit}^{2} .$$
(69)

 $^{^{26}}$ Note that the weights of the factor portfolios do not sum to one: the eigenvectors are normalized so that the sum of the squared elements is one.

Second, since the factor portfolio returns are orthogonal to each other, the covariance between q_{t+1}^{\star} and any risk variable $y_{k,t+1}$ can be written as a linear combination of the sum of the covariances between each of the factor portfolio returns and $y_{k,t+1}$:

$$\lambda_{kt} = -\sum_{i=1}^{N} \frac{\text{Cov}_t(r_{pi,t+1}, y_{k,t+1})}{\sqrt{v_{iit}}} S_{pit} .$$
(70)

This expression allows for a breakdown of the risk premium on an observable risk variable into the components due to the different factor portfolios.

Third, the ability of a subset of the factor portfolios to price all assets can be tested in the same way as any candidate pricing kernel.

B. Observable Factors: Multi-beta Models

A multi-beta model implies that expected excess returns are linear in the sensitivities of the returns to the risk variables, with coefficients given by the risk premia associated with the factors:

$$\mathbf{E}_t(\mathbf{r}_{t+1}) - r_{ft}\mathbf{1} = \boldsymbol{\beta}_t \boldsymbol{\lambda}_t , \qquad (71)$$

where $\boldsymbol{\beta}_t$ is an $N \times K$ matrix of projection coefficients of the returns on the risk variables. Hence, excess returns are described by the model

$$\mathbf{r}_{t+1} - r_{ft}\mathbf{1} = \boldsymbol{\beta}_t \boldsymbol{\lambda}_t + \boldsymbol{\beta}_t \mathbf{y}_{t+1} + \mathbf{e}_{t+1} , \qquad (72)$$

where \mathbf{e}_{t+1} is a vector of $N \times 1$ mean-zero perturbances orthogonal to the risk variables \mathbf{y}_{t+1} , with covariance matrix $\boldsymbol{\Sigma}_{eet}$.

Consider now the risk premia assigned by the minimum-variance kernel q_{t+1}^{\star} . From (16) I have

$$\lambda_t^{\star} = \alpha_{yt} \beta_t \lambda_t$$
$$= \Sigma_{yrt} \Sigma_{rrt}^{-1} \Sigma_{ryt} \Sigma_{yyt}^{-1} \lambda_t .$$
(73)

In the special case where $\Sigma_{yyt} = \mathbf{I}$, i.e., the factors are orthogonal with unit variance, the equation above simplifies to

$$\boldsymbol{\lambda}_t^{\star} = \boldsymbol{\Sigma}_{yrt} \boldsymbol{\Sigma}_{rrt}^{-1} \boldsymbol{\Sigma}_{ryt} \boldsymbol{\lambda}_t \;. \tag{74}$$

It is straightforward to verify that the $\Sigma_{yrt}\Sigma_{rrt}^{-1}\Sigma_{ryt}$ matrix equals the covariance matrix of the projections of the risk variables onto the span of asset returns, $\Sigma_{y^{\star}y^{\star}t}$,²⁷ i.e., $\lambda_t^{\star} = \Sigma_{y^{\star}y^{\star}t}\lambda_t$. Hence, the risk premia assigned by q_{t+1}^{\star} are linear combinations of the multi-beta premia through coefficients, which depend on the covariance matrix of the hedging-portfolio cash flows. In the two-factor case, for example,

$$\lambda_{1t}^{\star} = \sigma_{y^{\star}1t}^2 \lambda_{1t} + \sigma_{y^{\star}12t} \lambda_{2t} , \qquad (75)$$

where $\sigma_{y_{1t}}^2$ is also the R^2 of the projection of the first risk variable on the span of returns. In the special case where y_{1t} is perfectly tracked by the hedging portfolio, $\sigma_{y^*1t}^2 = 1$, $\sigma_{y^*12t} = 0$, and $\lambda_{1t}^* = \lambda_{1t}$.

Appendix B

Equation (32) combines Merton's (1973) intertemporal CAPM with Adler and Dumas (1983) international asset pricing model in presence of deviations from PPP. The continuous-time portfolio selection problem of a representative investor can be stated as follows:²⁸

$$\operatorname{Max} \ E \int_{t}^{T} V(C, P, s) \ ds , \qquad (76)$$

where $C = C(W, P, y_k, t)$ denotes nominal consumption expenditures, P is the price level index, V is a function homogeneous of degree zero in C and P expressing the instantaneous rate of indirect utility, and y_k is a state variable that affects utility through nominal consumption. Following Merton (1969), the wealth dynamics can be written as

$$dW = \left[\sum_{i=1}^{N} w_i(\mu_i - r_f) + r_f\right] W dt - C dt - \sum_{i=1}^{N} w_i \sigma_i dz_i , \qquad (77)$$

²⁷I have $\operatorname{Var}_{t}(\mathbf{y}_{t+1}^{?}) = \operatorname{Var}_{t}(\boldsymbol{\alpha}_{yt}\mathbf{r}_{t+1}) = \boldsymbol{\Sigma}_{yrt}\boldsymbol{\Sigma}_{rrt}^{-1}\boldsymbol{\Sigma}_{rrt}\boldsymbol{\Sigma}_{rrt}\mathbf{\Sigma}_{ryt} = \boldsymbol{\Sigma}_{yrt}\boldsymbol{\Sigma}_{rrt}^{-1}\boldsymbol{\Sigma}_{ryt}$.

²⁸See Appendix in Adler and Dumas (1983) for a detailed explanation of the necessary assumptions.

where $\mathbf{w} = \{w_i\}$ is $(N + 1) \times 1$ vector of weights, μ_i is the instantaneous expected nominal rate of return on security *i* expressed in a reference currency, σ_i is the instantaneous standard deviation of the nominal rate of return on security *i*, r_f is the risk-free rate expressed in a reference currency, and dz_i is the white noise of a standard Wiener process. Denoting with $J(W, P, y_k, t)$ the maximum value of (76) subject to (77), the Bellman principle states that total expected rate of increase of this function must be identically zero, so that

$$0 = \frac{Max}{(C, \mathbf{w})} \{ V(C, P, y_k, t) + J_t + J_W [-C + W(\sum_{i=0}^N w_i(\mu_i - r_f) + r_f)] + J_{y_k} \alpha + J_P P \pi + \frac{1}{2} J_{WW} W^2 \sum_{i=1}^N \sum_{j=1}^N w_i w_j \sigma_{ij} + \frac{1}{2} J_{yy} s^2 + \frac{1}{2} J_{PP} \sigma_{\pi}^2 P^2 + J_{WP} W P \sum_{i=1}^N w_i \sigma_{i\pi} + J_{y_k W} W \sum_{i=1}^N w_i \sigma_{iy_k} + J_{y_k P} \sigma_{y_k \pi} \},$$
(78)

where π is the inflation rate in each country expressed in local units, σ_{ij} are the instantaneous covariances of the nominal rates of return on the various securities, σ_{π}^2 is the instantaneous variance of the inflation rate, α is the mean value of the state variable y_k , $\sigma_{i\pi}$ is the covariance between security i and the inflation rate π , and $\sigma_{y_k\pi}$ is the covariance between the state variable y_k and the inflation rate π .²⁹ Moreover, the homogeneity of degree zero of the function V implies that $J(W, P, y_k, t)$ and $C(W, P, y_k, t)$ that satisfy (78) must be homogeneous of degree zero in W and P: $J_P \equiv -(W/P)J_W$, $J_{PW} \equiv (-1/P)J_W - (W/P)J_{WW}$, $J_{PP} = 2(W/P^2)J_W + (W/P)^2J_{WW}$. Hence, (78) can be rewritten as

$$0 = \frac{Max}{(C, \mathbf{w})} \{ V(C, P, y_k, t) + J_t + J_W [-C + W(\sum_{i=0}^N w_i(\mu_i - r_f) + r_f)] + J_{y_k}\alpha - W\pi J_W + \frac{1}{2} J_{WW} W^2 \sum_{i=1}^N \sum_{j=1}^N w_i w_j \sigma_{ij} + \frac{1}{2} J_{yy} s^2 + W J_W \sigma_\pi^2 + \frac{1}{2} \sigma_\pi^2 W^2 J_{WW} - J_W W \sum_{i=1}^N w_i \sigma_{i\pi} - W^2 J_{WW} \sum_{i=1}^N w_i \sigma_{i\pi} + J_{y_k W} W \sum_{i=1}^N w_i \sigma_{iy_k} + J_{y_k P} \sigma_{y_k \pi} \}.$$
(79)

Taking the first order conditions of (79) with respect to C and \mathbf{w} , I obtain

$$V_C = J_W , \text{ and}$$
(80)

$$0 = J_W(\mu_i - r_f) + W J_{WW} \sum_{j=1}^N w_j \sigma_{ij} - J_W \sigma_{i\pi} - W J_{WW} \sigma_{i\pi} + J_{y_k W} \sigma_{iy_k} .$$
(81)

²⁹See Chapter 13 of Ingersoll (1987) for a definition of the dynamics of the state variables.

Solving (79) for the optimal portfolio of risky assets of investor l directly in vector notation and reintroducing the time dimension, I obtain (32)

$$\mathbf{w}_t^l = \alpha^l \boldsymbol{\Sigma}_{rrt}^{-1} [E_t(\mathbf{r}_{t+1}) - r_{ft} \mathbf{1}] + (1 - \alpha^l) \boldsymbol{\Sigma}_{rrt}^{-1} \mathbf{s}_{r\pi t}^l + \beta^l \boldsymbol{\Sigma}_{rrt}^{-1} \mathbf{s}_{ry_k t}^l , \qquad (82)$$

where $\mathbf{s}_{ry_k t}^l$ and $\mathbf{s}_{r\pi t}^l$ are the vectors of time-varying covariances between each country's security returns and the k-th state variable and level of inflation respectively. Moreover, I define $\alpha^l \equiv -\frac{J_{Wt}^l}{J_{WWt}^l W_t^l}$ and $\beta^l \equiv (\alpha^l)^2 \times \frac{J_{Wyk}^l}{W_t^l}$.

Appendix C

While the minimum-variance pricing kernel, q_{t+1}^{\star} , satisfies the law of one price (equation (1)), in general it does not satisfy the no-arbitrage condition, $q_{t+1}^{\star} > 0$. Nonetheless, as in HJ, I can extend the analysis to take this restriction into account.

Let $\tilde{\boldsymbol{\alpha}}_t$ denote an $N \times 1$ coefficient vector, and define $\tilde{q}_{t+1} \equiv [1 - (\mathbf{r}_{t+1} - \mathbf{E}_t(\mathbf{r}_{t+1}))^\top \tilde{\boldsymbol{\alpha}}_t]^+ \equiv \max\{1 - (\mathbf{r}_{t+1} - \mathbf{E}_t(\mathbf{r}_{t+1}))^\top \tilde{\boldsymbol{\alpha}}_t, 0\}$. Assume

$$\mathbf{E}_t(\tilde{q}_{t+1}\mathbf{r}_{t+1}) = r_{ft}\mathbf{1} . \tag{83}$$

The random variable \tilde{q}_{t+1} has the smallest variance among all nonnegative random variables satisfying restriction (83).

Consider the risk premium $\tilde{\lambda}_{kt}$ assigned by \tilde{q} . I can write

$$\begin{split} \tilde{\lambda}_{kt} &\equiv -\mathbf{E}_t[(\tilde{q}_{t+1}-1)y_{k,t+1}] \\ &= -\mathbf{E}_t[(\tilde{q}_{t+1}-1)y_{k,t+1}^{\star}] - \mathbf{E}_t[(\tilde{q}_{t+1}-1)(y_{k,t+1}-y_{k,t+1}^{\star})]\tilde{\lambda}_{kt} \\ &= \lambda_{kt}^{\star} - \mathbf{E}_t[(\tilde{q}_{t+1}-1)(y_{k,t+1}-y_{k,t+1}^{\star})] , \end{split}$$
(84)

where, in general, $E_t[(\tilde{q}_{t+1}-1)(y_{k,t+1}-y_{k,t+1}^*)] \neq 0$. Hence, when the positivity restriction is imposed, the risk premium assigned by the minimum-variance kernel differs from the mean cash flow generated by the hedging portfolio by the quantity $-E_t[(\tilde{q}_{t+1}-1)(y_{k,t+1}-y_{k,t+1}^*)]$. If \tilde{q}_{t+1} is volatile, and if $y_{k,t+1}$ is mimicked poorly by its nearest hedge, then there is the potential for the discrepancy to be substantial.

References

- Adler, Michael, and Bernard Dumas, 1983, International portfolio choice and corporation finance: A synthesis, Journal of Finance 38, 925-984.
- [2] Balduzzi, Pierluigi, and Hèdi Kallal, 1997, Risk premia and variance bounds, Journal of Finance 52, 1913-1949.
- [3] Balduzzi, Pierluigi, and Cesare Robotti, 2001, Minimum-variance kernels, economic risk premia, and tests of multi-beta models, Boston College, Unpublished Manuscript.
- Bekaert, Geert, and Campbell R. Harvey, 1995, Time-varying world market integration, Journal of Finance 50, 403-444.
- [5] Campbell, John Y., Understanding risk and return, 1996, Journal of Political Economy 104, 298-345.
- [6] Chen, Nai-Fu, Richard Roll, and Stephen A. Ross, 1986, Economic forces and the stock market, Journal of Business 59, 383-403.
- [7] De Santis, Giorgio, and Bruno Gerard, 1997, International asset pricing and portfolio diversification with time-varying risk, Journal of Finance 52, 1881-1912.
- [8] De Santis, Giorgio, and Bruno Gerard, 1998, How big is the premium for currency risk?, Journal of Financial Economics 49, 375-412.
- [9] Dumas, Bernard, and Bruno Solnik, 1995, The world price of foreign exchange risk, Journal of Finance 50, 445-479.
- [10] Ferson, Wayne E., and Campbell R. Harvey, 1991, The variation of economic risk premiums, Journal of Political Economy 99, 385-415.
- [11] Ferson, Wayne E., and Campbell R. Harvey, 1993, The risk and predictability of international equity returns, Review of Financial Studies 6, 527-566.

- [12] Grauer, Frederick L.A., Robert H. Litzenberger, and Richard E. Stehle, 1976, Sharing rules and equilibrium in an international capital market under uncertainty, Journal of Financial Economics 3, 233-256.
- [13] Hansen, Lars P., and Ravi Jagannathan, 1991, Implications of security market data for models of dynamic economies, Journal of Political Economy 99, 225-262.
- [14] Hansen, Lars P., and Ravi Jagannathan, 1997, Assessing specification errors in stochastic discount factors models, Journal of Finance 52, 557-590.
- [15] Hansen, Lars P., and Kenneth J. Singleton, 1982, Generalized instrumental variables estimation of nonlinear rational expectations models, Econometrica 50, 1269-1286.
- [16] Harrison, Michael J., and David M. Kreps, 1979, Martingales and arbitrage in multiperiod securities markets, Journal of Economic Theory 20, 381-408.
- [17] Hodrick, Robert J., 1981, International asset pricing with time-varying risk premia, Journal of International Economics 11, 573-587.
- [18] Hodrick, Robert J., David Tat-Chee Ng, and Paul Sengmüller, 1999, An international dynamic asset pricing model, Columbia University, Working Paper.
- [19] Ingersoll, Jonathan E., 1987, Theory of financial decision making, Rowman&Littlefield, Totowa, NJ.
- [20] Lamont, Owen, 2001, Economic Tracking Portfolios, forthcoming Journal of Econometrics.
- [21] Lintner, John, 1965, Security prices, risk and maximal gains from diversification, Journal of Finance 20, 587-615.
- [22] Merton, Robert C., An intertemporal capital asset pricing model, 1973, Econometrica 41, 867-887.
- [23] Mossin, Jan, 1966, Equilibrium in a capital asset market, Econometrica 35, 768-783.

- [24] Sercu, Piet, 1980, A generalization of the international asset pricing model, Reveu de l' Association Francaise de Finance 1, 91-135.
- [25] Sharpe, William F., 1964, Capital asset prices: A theory of market equilibrium under conditions of risk, Journal of Finance 19, 425-442.
- [26] Solnik, Bruno H., 1974, An international market model of security price behavior, Journal of Financial and Quantitative Analysis 9, 537-554.
- [27] Stehle, Richard, 1977, An empirical test of the alternative hypotheses of national and international pricing of risky assets, Journal of Finance 32, 493-502.
- [28] Vassalou, Maria, 2000, Exchange rate and foreign inflation risk premiums in global equity returns, manuscript, Journal of International Money and Finance 19, 433-470.
- [29] Zhang, Xiaoyan, 2001, Specification Tests of Asset Pricing Models in International Market, Unpublished Manuscript, Columbia University.

Table ISummary Statistics of Asset Returns

I report summary statistics of monthly returns on the equity indices of four countries from MSCI. Indices include dividends. The sample covers the period April 1970 through October 1998 (343 observations). All returns are in percentage points per month and are denominated in U.S. dollars. "Corr_{τ}" denotes the autocorrelation coefficient of order τ . "Q₃₆" denotes the Ljung–Box Q statistics of order 36 (p–values in parenthesis).

Std. Dev. $\overline{\mathrm{Corr}_4}$ Mean Corr_1 Corr_2 Corr_3 Corr_{24} Corr_{36} Country $Corr_{12}$ Q_{36} 27.586 United States 1.11134.4171 0.002 -0.034 0.007 -0.018 0.048 0.005 -0.033 (0.842)48.085 -0.026 United Kingdom 1.33047.05250.085-0.101 0.0520.013 -0.023 0.047 (0.086)46.582 Japan 1.22966.67570.091-0.022 0.0780.043 0.032 0.006 0.033 (0.111) 41.417 -0.017 -0.013 0.029 Germany 1.20995.90690.0520.067 -0.017 0.022 (0.246)

Panel A: Means, Standard Deviations, and Autocorrelations

Panel B: Correlation Matrix

Country	United States	United Kingdom	Japan	Germany
United States	1.000	0.505	0.264	0.367
United Kingdom		1.000	0.358	0.425
Japan			1.000	0.366
Germany				1.000

Table II Summary Statistics of Economic Variables

I report summary statistics of economic risk factors for the United States, the United Kingdom, Japan and Germany. The sample covers the period April 1970 through October 1998 (343 observations). *INFL* denotes the unexpected rate of inflation (percentage points per month). *SPOT* denotes the change in the logarithm of the spot exchange rate (percentage points per month). *WLDMK* denotes the rate of return on the world market index from MSCI (percentage points per month). "Corr_{τ}" denotes the autocorrelation coefficient of order τ . "Q₃₆" denotes the Ljung–Box Q statistics of order 36 (p–values in parenthesis).

Variable	Mean	Std. Dev.	Corr_1	Corr_2	Corr_3	Corr_4	Corr_{12}	Corr_{24}	Corr_{36}	Q_{36}
$INFL_{US}$	-0.0002	0.2406	0.160	-0.023	-0.112	-0.154	0.167	0.093	0.138	95.189 (0.000)
$INFL_{UK}$	-0.0001	0.6130	0.164	-0.019	-0.061	-0.119	0.475	0.418	0.387	295.03 (0.000)
$INFL_{JAP}$	0.0004	0.6770	0.100	-0.220	-0.060	-0.013	0.439	0.397	0.378	373.09 (0.000)
$INFL_{GER}$	-0.0007	0.3129	0.188	0.051	-0.019	-0.082	0.299	0.094	0.165	134.34 (0.000) 33.924
$SPOT_{S/GBP}$	0.1049	3.0593	0.086	0.025	-0.019	0.012	-0.011	-0.038	-0.023	33.924 (0.568) 37.342
$SPOT_{YEN}$	-0.3288	3.3611	0.080	0.033	0.048	0.039	0.073	-0.045	-0.085	37.342 (0.407)
$SPOT_{/DM}$	-0.2312	3.2740	0.036	0.070	0.014	-0.005	-0.007	0.004	0.017	32.18Ó (0.651)
WLDMK	1.0551	4.1392	0.068	-0.055	0.013	-0.021	0.053	0.045	-0.019	45.179 (0.140)

Panel A: Means, Standard Deviations, and Autocorrelations

Panel B: Correlation Matrix

Variable	$INFL_{US}$	$INFL_{UK}$	$INFL_{JAP}$	$INFL_{GER}$	$SPOT_{S/GBP}$	$SPOT_{YEN}$	$SPOT_{M}$	WLDMK
$INFL_{US}$	1.000	0.057	0.141	0.157	0.016	0.013	0.095	-0.196
$INFL_{UK}$		1.000	0.337	0.224	-0.089	-0.015	-0.081	0.110
$INFL_{JAP}$			1.000	0.082	-0.009	0.029	-0.014	-0.008
$INFL_{GER}$				1.000	0.033	0.082	0.097	0.019
$SPOT_{S/GBP}$					1.000	0.472	0.650	-0.261
$SPOT_{YEN}$						1.000	0.578	-0.310
$SPOT_{M}$							1.000	-0.228
WLDMK								1.000

Table III Summary Statistics of Instrumental Variables

I report summary statistics of the intruments used in the analysis. The sample covers the period March 1970 through September 1998 (343 observations). DINFLUS denotes the lagged change in the U.S. rate of inflation. EURO denotes the Eurodollar deposit rate. USDIVYLD denotes the U.S. dividend yield in excess of the Eurodollar rate. WLDMK denotes the world rate of return. DEFPREM represents the U.S. default premium. All variables are expressed in percentage points per month. "Corr_{τ}" denotes the autocorrelation coefficient of order τ . "Q₃₆" denotes the Ljung–Box Q statistics of order 36 (p-values in parenthesis).

Panel A: Means, Standard Deviations, and Autocorrelations

Variable	Mean	Std. Dev.	Corr_1	Corr_2	Corr_3	Corr_4	Corr_{12}	Corr_{24}	Corr_{36}	Q_{36}
DINFLUS	-0.0012	0.2719	-0.355	-0.054	-0.033	-0.077	0.142	0.157	0.169	140.350 (0.000) 4051.0
EURO	0.6503	0.2780	0.967	0.920	0.879	0.844	0.667	0.343	0.151	4051.0 (0.000) 2768.6
USDIVYLD	-0.3353	0.2172	0.950	0.880	0.822	0.771	0.543	0.150	-0.088	
WLDMK	1.1672	4.1166	0.068	-0.035	0.015	-0.024	0.062	0.046	-0.016	(0.000) 46.196 (0.119)
DEFPREM	0.0141	1.1768	-0.198	-0.036	-0.005	0.013	-0.067	-0.054	0.028	(0.119) 65.046 (0.002)

Panel B: Correlation Matrix

Variable	DINFLUS	EURO	USDIVYLD	WLDMK	DEFPREM
DINFLUS	1.000	-0.008	0.000	-0.075	0.098
EURO		1.000	-0.952	-0.126	-0.046
USDIVYLD			1.000	0.131	0.077
WLDMK				1.000	-0.003
DEFPREM					1.000

Table IV Instruments Selection and Heteroskedasticity T-statistics, in parentheses, are adjusted for heteroskedasticity and serial correlation.

Var CONST JAN DINFLUS EURO USDIVYLD WLDMK DEFPREM 0.971 1.716 -0.4081.266 1.605 -0.101 0.231 United States (4.099) (1.696) 3.114 (1.393) (1.453) 2.828 (2.468) (1.893) 3.408 (0.882) 0.084 (-1.759) -0.469 (-0.349) 0.086 United Kingdom (2.976) (-1.294)(2.964)(0.194) (0.150) 0.448 -0.7722.273 (2.137) 2.835 0.629 0.154 Japan (3.236)(0.371)(-2.229)(2.677)(1.508)(0.392) -0.659 (-2.051) 1.243 (3.781) -0.410 0.457 0.863 0.161 0.574 Germany (-0.373)(0.476) (0.898) (0.434)(1.475)

Panel A: Slope Estimates of Mean Equations

Panel B: Slope Estimates of Variance Equations

Var	CONST	JAN	DINFLUS	EURO	USDIVYLD	WLDMK	DEFPREM
United States	3.192	1.062	0.105	0.039	-0.224	-0.472	-0.359
	(21.212)	(1.739)	(0.840)	(0.070)	(-0.414)	(-2.739)	(-2.438)
United Kingdom	4.809	1.521	0.098	1.587	1.362	-0.079	-0.375
	(20.170)	(0.847)	(0.427)	(1.848)	(1.578)	(-0.256)	(-0.887)
Japan	5.083	-0.528	-0.364	-1.262	_1.194	-0.016	-0.056
	(22.131)	(-0.709)	(-1.595)	(-1.956)	(_1.867)	(-0.057)	(-0.221)
Germany	4.458	-0.252	0.048	-0.825	-1.028	-0.216	0.106
	(21.111)	(-0.351)	(0.238)	(-1.361)	(-1.712)	(-0.932)	(0.423)

Panel C: Average Slope Estimates and Differences in Slope Estimates

Var	DINFLUS	EURO	USDIVYLD	WLDMK	DEFPREM
$\bar{\mu}_r$	-0.577	1.706	2.178	0.194	0.261
	(-2.441)	(2.360)	(3.051)	(0.712)	(0.876)
\bar{v}_r	-0.028	-0.115	-0.271	_0.196	_0.171
	(-0.222)	(0.263)	(-0.630)	(_1.181)	(_0.967)
$\bar{\mu}_r - \bar{v}_r$	-0.549	1.821	2.449	0.390	0.432
	(-1.972)	(2.095)	(2.855)	(1.361)	(1.499)

I report coefficients of the economic risk premia on the following eight economic factors: $INFL_{US}$, $INFL_{UK}$, $INFL_{JAP}$, $INFL_{GER}$, $SPOT_{AP}$, $SPOT_{DM}$, WLDMK. T-statistics, in parentheses, are adjusted for heteroskedasticity and serial correlation. The t-statistics refer to the case where the composition of q^* is estimated outside and inside of the GMM algorithm, respectively. This table presents estimates of the conditional premia when the set of returns is augmented to include managed portfolios. In this case, note that the intercept can be interpreted as the unconditional risk premium with the larger set of securities.

Conditional Risk Premia

Risk Premia	$INFL_{US}$	$INFL_{UK}$	$INFL_{JAP}$	$INFL_{GER}$	$SPOT_{S/GBP}$	$SPOT_{\N VEN}$	$SPOT_{\DM}$	WLDMK
CONST	-0.0128	-0.0198	-0.0145	0.0102	-0.0317	-0.0233	-0.0275	0.1171
	(-0.61)	(-0.94)	(-0.73)	(0.45)	(-1.52)	(-1.10)	(-1.12)	(3.52)
	(-0.45)	(-0.85)	(-0.62)	(0.48)	(-0.95)	(-0.63)	(-0.75)	(2.28)
JAN	-0.0680	0.0329	0.0258	0.0276	-0.0035	-0.0016	-0.0222	0.0895
	(-1.84)	(0.81)	(1.16)	(0.64)	(-0.17)	(-0.11)	(-0.87)	(1.59)
	(-1.54)	(1.11)	(0.93)	(1.15)	(-0.12)	(-0.05)	(-0.57)	(1.56)
DINFLUS	0.0077	0.0069	0.0432	0.0565	0.0203	0.0381	0.0457	-0.1249
	(0.26)	(0.30)	(1.49)	(1.59)	(0.89)	(1.41)	(1.28)	(-2.37)
	(0.17)	(0.20)	(1.06)	(1.43)	(0.74)	(1.22)	(1.35)	(-2.20)
EURO	-0.0149	0.0231	0.0313	-0.0948	-0.0666	-0.3286	-0.1126	0.2888
	(-0.23)	(0.34)	(0.53)	(-1.25)	(-0.98)	(-3.45)	(-1.51)	(2.65)
	(-0.18)	(0.35)	(0.41)	(-1.13)	(-0.72)	(-2.09)	(-1.02)	(1.67)
USDIVYLD	-0.0570	0.1027	0.0433	-0.0973	-0.1472	-0.3706	-0.1827	0.4617
	(-0.89)	(1.48)	(0.66)	(-1.11)	(-1.92)	(-3.80)	(-2.16)	(3.80)
	(-0.72)	(1.48)	(0.52)	(-1.04)	(-1.50)	(-2.44)	(-1.56)	(2.73)
WLDMK	-0.0280	0.0189	-0.0542	-0.0062	-0.0416	-0.0449	-0.0420	0.0331
	(-1.07)	(0.64)	(-1.58)	(-0.35)	(-1.42)	(-1.78)	(-1.31)	(0.87)
	(-0.88)	(0.67)	(-1.57)	(-0.23)	(-1.36)	(-1.20)	(-1.20)	(0.60)
DEFPREM	0.0108	-0.0254	-0.0508	-0.0344	-0.0201	-0.0589	-0.0708	0.0527
	(0.32)	(-0.70)	(-1.60)	(-0.84)	(-0.78)	(-1.88)	(-1.66)	(0.82)
	(0.29)	(-0.84)	(-1.74)	(-0.70)	(-0.60)	(-1.34)	(-1.39)	(0.91)

I report coefficients of the economic risk premia on the following eight economic factors: $INFL_{US}$, $INFL_{UK}$, $INFL_{JAP}$, $INFL_{GER}$, $SPOT_{AP}$, $SPOT_{DM}$, WLDMK. T-statistics, in parentheses, are adjusted for heteroskedasticity and serial correlation. The t-statistics refer to the case where the composition of q^* is estimated outside and inside of the GMM algorithm, respectively. This table presents estimates of the conditional premia when the set of returns is augmented to include managed portfolios. In this case, note that the intercept can be interpreted as the unconditional risk premium with the larger set of securities.

Conditional Risk Premia

Risk Premia	$INFL_{US}$	$INFL_{UK}$	$INFL_{JAP}$	$INFL_{GER}$	$SPOT_{S/GBP}$	$SPOT_{\N VEN}$	$SPOT_{\DM}$	WLDMK
CONST	-0.0128	-0.0198	-0.0145	0.0102	-0.0317	-0.0233	-0.0275	0.1171
	(-0.57)	(-1.05)	(-0.98)	(0.85)	(-0.95)	(-0.65)	(-0.81)	(2.28)
	(-0.45)	(-0.85)	(-0.62)	(0.48)	(-0.95)	(-0.63)	(-0.75)	(2.28)
JAN	-0.0198	0.0244	-0.0252	0.0300	0.0108	0.0055	-0.0013	0.0859
	(-0.59)	(0.68)	(-1.40)	(1.79)	(0.34)	(0.20)	(-0.04)	(1.51)
	(-0.46)	(0.65)	(-0.97)	(1.24)	(0.31)	(0.17)	(-0.03)	(1.49)
DINFLUS	0.0361	-0.0358	0.0075	-0.0102	0.0493	0.0931	0.0830	-0.1249
	(1.09)	(-1.32)	(0.50)	(-0.74)	(1.51)	(2.41)	(2.55)	(-2.24)
	(0.92)	(-1.16)	(0.33)	(-0.47)	(1.33)	(2.20)	(2.30)	(-2.23)
EURO	-0.0805	0.0828	0.0320	0.0310	-0.2296	-0.2157	-0.1636	0.3384
	(-1.15)	(1.22)	(0.62)	(0.80)	(-2.31)	(-1.92)	(-1.58)	(1.97)
	(-0.89)	(1.01)	(0.42)	(0.47)	(-2.21)	(-1.92)	(-1.52)	(1.97)
USDIVYLD	-0.1200	0.1027	0.0491	0.0091	-0.2806	-0.2693	-0.2259	0.5189
	(-1.72)	(1.52)	(0.90)	(0.23)	(-2.79)	(-2.41)	(-2.22)	(3.07)
	(-1.35)	(1.24)	(0.61)	(0.15)	(-2.72)	(-2.43)	(-2.11)	(3.07)
WLDMK	0.0089	0.0298	-0.0007	0.0097	-0.0501	-0.0914	-0.0444	0.0364
	(0.31)	(1.18)	(-0.03)	(0.68)	(-1.23)	(-1.84)	(-1.06)	(0.64)
	(0.27)	(1.06)	(-0.02)	(0.52)	(-1.28)	(-1.92)	(-1.06)	(0.63)
DEFPREM	-0.06/4	-0.0076	-0.0231	-0.0184	-0.0439	-0.0358	-0.06/5	0.0600
	(-1.83)	(-0.27)	(-1.22)	(-0.98)	(-0.96)	(-0.72)	(-1.28)	(1.00)
	(-1.52)	(-0.22)	(-0.80)	(-0.68)	(-0.93)	(-0.73)	(-1.26)	(1.00)

Table VII Unconditional Risk Premia, Volatility of Hedging Portfolios and Sharpe Ratios

I report unconditional economic risk premia (λ_0), volatilities of the mimicking portfolios' excess cash flows (v_0) and Sharpe ratios ($S_{y_k^2}$) commanded by the following economic factors: $INFL_{US}$, $INFL_{UK}$, $INFL_{JAP}$, $INFL_{GER}$, $SPOT_{S/GBP}$, $SPOT_{S/YEN}$, $SPOT_{MK}$. T-statistics, in parentheses, are adjusted for heteroskedasticity and serial correlation. The t-statistics refer to the case where the composition of y^* is estimated outside the GMM algorithm.

Sharpe Ratios	λ_0	v_0	$S_{y_{k}^?}$
$INFL_{US}$	-0.0128	0.4279	-0.0301
I I V I L US	(-0.56)	(16.06)	(-0.56)
$INFL_{UK}$	-0.0198	0.3563	-0.0558
I I V I D U K	(-1.03)	(9.61)	(-0.98)
$INFL_{JAP}$	-0.0145	0.2782	-0.0522
	(-0.97)	(17.60)	(-0.97)
$INFL_{GER}$	0.0102	0.2262	0.0450
	(0.84)	(16.17)	(0.85)
$SPOT_{S/GBP}$	-0.0317	0.6336	-0.0501
	(-0.93)	(21.75)	(-0.93)
$SPOT_{YEN}$	-0.0233	0.6846	-0.0339
-	(-0.63)	(20.55)	(-0.63)
$SPOT_{M}$	-0.0275	0.6447	-0.0427
OI OI \$/DM	(-0.79)	(18.63)	(-0.79)
WLDMK	0.1170	0.9947	0.1177
	(2.18)	(18.93)	(2.10)

Table VIII Tests of IS-CAPM, I-CAPM and II-CAPM

I perform conditional tests of the International Static CAPM (IS-CAPM), the International CAPM in presence of deviations from PPP (I-CAPM (PPP)), the International CAPM in presence of currency risk (I-CAPM (SPOT)) and the international intertemporal CAPM (II-CAPM(PPP) and II-CAPM(SPOT)) by using region subset tests (χ^2), Hansen-Jagannathan variance bounds (HJV), and Hansen-Jagannathan distance measures (HJD), respectively. The benchmark standard deviation of the scaled unrestricted normalized minimum-variance kernel is 0.3865. The standard deviations of the restricted normalized minimum-variance kernels with conditional information are reported in column two. With q_{IS}^* , q_{I-PPP}^* , q_{I-SPOT}^* , q_{II-PPP}^* , and $q_{II-SPOT}^*$ I denote the normalized minimum-variance kernels with conditional information for the IS-CAPM, I-CAPM (PPP), I-CAPM(SPOT), II-CAPM (PPP), and II-CAPM(SPOT), respectively.

Models		With conditioning information					
Statistics	$\begin{array}{c} \chi^2_{(dof)} \\ (p-value) \end{array}$	$\begin{array}{c} HJV\\ (t-stat.)\\ std(q_r^{\star}) \ r=IS,I-PPP,I-SPOT,II-PPP,II-SPOT\end{array}$	HJD (t-stat.)				
IS-CAPM	$\begin{array}{c} 33.09_{(27)} \\ (0.19) \end{array}$	$-0.2630 \ (-10.68) \ std(q_{IS}^{T}) = 0.1231 \ -0.2038$	$0.3659 \\ (12.62)$				
I-CAPM(PPP)	$\begin{array}{c} 31.16_{(23)} \\ (0.12) \end{array}$	$^{(-7.48)}_{std(q^{\star}_{I-PPP})=0.1825}$	$0.3402 \\ (12.47)$				
I-CAPM (SPOT)	$\begin{array}{c} 31.17_{(24)} \\ (0.14) \end{array}$	$\begin{array}{c} -0.2595\\ (-10.46)\\ std(q_{I-SPOT}^{*})=0.1266\\ -0.1826\end{array}$	$\begin{array}{c} 0.3646 \\ (12.43) \end{array}$				
II-CAPM(PPP)	$\begin{array}{c} 30.37_{(19)} \\ (0.05) \end{array}$	$\begin{array}{r} -0.1826 \\ (-7.26) \\ std(q_{II-PPP}^{\star}) = 0.2037 \\ -0.2549 \end{array}$	$\begin{array}{c} 0.3280 \\ (12.28) \end{array}$				
II-CAPM(SPOT)	$\begin{array}{c} 30.42_{(20)} \\ (0.06) \end{array}$	$^{-0.2549}_{(-10.21)}$ $std(q^{\star}_{II-SPOT})=0.1313$	$\begin{array}{c} 0.3630 \\ (12.35) \end{array}$				

48

Table IX Differences in Hansen-Jagannathan Variance Bounds and Distance Measures

I test whether the pricing implications delivered by the International Static CAPM (IS-CAPM), the International CAPM in presence of deviations from PPP (I-CAPM (PPP)), the International CAPM in presence of currency risk (I-CAPM (SPOT)) and the international intertemporal CAPM (II-CAPM(PPP) and II-CAPM(SPOT)) are statistically different from each other. The estimates of the differences in HJV and HJD measures are obtained by exactly identified GMM. T-statistics are obtained using the delta method.

Statistics	$\begin{array}{c} (HJV_r-HJV_s)\\ (t-stat.)\\ std(q_r^2) r,s=IS,I-PPP,I-SPOT,II-PPP,II-SPOT,r\neq s\end{array}$	$(HJD_{r}-HJD_{s})$ (t-stat.)
IS-CAPM – I-CAPM(PPP)	0.0592 (6.92)	-0.0256 (-2.26)
IS-CAPM – I-CAPM(SPOT)	0.0035 (1.85)	-0.0012 (-0.62)
IS-CAPM – II-CAPM(PPP)	0.0804 (9.23)	_0.0378 (_2.33)
IS–CAPM – II-CAPM(SPOT)	0.0081 (2.89)	_0.0028 (_0.79)
I-CAPM(PPP) – I-CAPM (SPOT)	0.0557 (6.32)	-0.0244 (-2.09)
I-CAPM(PPP) - II-CAPM(PPP)	0.0212 (2.89)	_0.0122 (_1.57)
I-CAPM(PPP) - II-CAPM(SPOT)	-0.0511 (-6.03)	0.0227 (2.00)
I-CAPM(SPOT) - II-CAPM(PPP)	0.0770 (8.52)	-0.0366 (-2.19)
I-CAPM(SPOT) - II-CAPM(SPOT)	0.0047 (2.20)	_0.0016 (_0.59)
II-CAPM(PPP) - II-CAPM(SPOT)	0.0723 (8.13)	_0.0350 (_2.11)

49

Table X Hedging Demands

I report the unconditional normalized hedging demands for global market, inflation, and foreign exchange risks. T-statistics, in parentheses, are adjusted for heteroskedasticity and serial correlation.

Hedging Demands	α
$SPOT_{S/GBP}$	0.129 (5.26)
$SPOT_{YEN}$	0.038 (1.76)
$SPOT_{M}$	0.058 (2.38)
WLDMK	0.109 (10.04)
$INFL_{US}$	0.222 (8.91)
$INFL_{UK}$	0.364 (12.32)
$INFL_{JAP}$	0.173 (4.86)
$INFL_{GER}$	-0.093 (-3.04)